# **UC Davis**

**Recent Work** 

### Title

Reflective Cracking Study: First-level Report on Laboratory Shear Testing

### Permalink

https://escholarship.org/uc/item/63h2v2td

### Authors

Tsai, Bor-Wen Jones, David Harvey, John T <u>et al.</u>

### **Publication Date**

2008-10-01

Peer reviewed

# Reflective Cracking Study: First-Level Report on Laboratory Shear Testing

Authors: I. Guada, J. Signore, B. Tsai, D. Jones, J. Harvey, and C. Monismith

Partnered Pavement Research Program (PPRC) Contract Strategic Plan Element 4.10: Development of Improved Rehabilitation Designs for Reflective Cracking

### **PREPARED FOR:**

California Department of Transportation Division of Research and Innovation Office of Roadway Research

### **PREPARED BY:**

University of California Pavement Research Center UC Davis, UC Berkeley





#### DOCUMENT RETRIEVAL PAGE

Research Report: UCPRC-RR-2006-11

Title: Reflective Cracking Study: First-level Report on Laboratory Shear Testing

Authors: I. Guada, J. Signore, B. Tsai, D. Jones, J. Harvey, and C. Monismith

<b>Prepared for:</b>	<b>FHWA No:</b>	Date:
Caltrans	CA091073L	September 2007
Contract No:	<b>Client Reference No:</b>	<b>Status:</b>
65A0172	SPE 4.10	Stage 6, Approved Version

#### Abstract:

This report contains a summary of the laboratory repeated load shear tests on mixes used as overlays on the Reflective Cracking Study Test Track at the Richmond Field Station. Evaluation of the results of the laboratory study on shear response of the overlay mixes reported herein included the effects of mix temperatures, air-void content, aging, mixing and compaction conditions, aggregate gradation, and shear stress level. Shear testing was performed to assess expected rutting performance at high temperatures. Mixes with five binders were tested, namely AR4000, asphalt rubber, and three modified binders termed MB4 (meeting the Caltrans MB4 specification [2003]), MB15 (meeting the MB4 specification and containing 15 percent recycled tire rubber, referred to as MAC15), and MAC15TR (Southern California GreenBook specification, containing 15 percent recycled tire rubber, referred to as MAC15). A full factorial considering all the variables required a total of 2,880 tests. This was reduced to 186 tests to accommodate time and fund constraints. Based on the shear test results for the mixes used in the overlay experiment, mix rankings for Cycles to 5 Percent Permanent Shear Strain, Permanent Shear Strain at 5,000 Cycles, and Resilient Shear Modulus ( $G^*$ ) were:

- Permanent Shear Strain at 5,000 Cycles (best performance to worst): AR4000-D; MAC15-G; RAC-G; MB4-G; MB15-G.
- Cycles to 5 Percent Permanent Shear Strain (best performance to worst): AR4000-D; MAC15-G; RAC-G; MB4-G; MB15-G.
- Resilient Shear Modulus (G\*) (stiffest to least stiff): AR4000-D; RAC-G; MAC15-G; MB4-G; MB15-G

The results show that the rankings of different mixes for expected rutting performance are the same for the two permanent deformation parameters calculated from Repeated Simple Shear Test (RSST) results: Permanent Shear Strain at 5,000 Cycles and Cycles to 5 Percent Permanent Shear Strain. The RSST demonstrated that the dense-graded mix with unmodified AR-4000 asphalt performed better than the gap-graded mixes with modified binders. Of the three gap-graded mixes with modified binders, the MAC15 mix exhibited the highest resistance to shear deformation while the MB15 mix exhibited the lowest. Dense-graded mixes with the modified binders (MB4, MB15, MAC15) generally showed an increase in permanent shear deformation resistance and shear stiffness compared to gap-graded mixes with the same binders. Until a range of pavement types and environments are evaluated in the second<sup>-</sup> level analysis, these results provide only a general indication of the relative performance of the modified binders with respect to rutting performance.

#### **Keywords:**

Shear testing, rutting, overlay, modified binder, HVS test, MB Road

**Proposals for implementation:** Implementation recommendations will be made after completion of other testing and analyses.

#### **Related documents:**

UCPRC-RR-2005-03, UCPRC-RR-2006-08

#### Signatures:

8				
I. Guada 1st Author	C Monismith Technical Review	D. Spinner Editor	J. Harvey Principal Investigator	M Samadian Caltrans Contract Manager

The contents of this report reflect the views of the authors who are responsible for the facts and accuracy of the data presented herein. The contents do not necessarily reflect the official views or policies of the State of California or the Federal Highway Administration. This report does not constitute a standard, specification, or regulation.

### **PROJECT OBJECTIVES**

The objective of this project is to develop improved rehabilitation designs for reflective cracking for California.

This objective will be met after completion of four tasks identified by the Caltrans/Industry Rubber Asphalt Concrete Task Group (RACTG):

- 1. Develop improved mechanistic models of reflective cracking in California,
- 2. Calibrate and verify these models using laboratory and HVS testing,
- 3. Evaluate the most effective strategies for reflective cracking, and
- 4. Provide recommendations for reflective cracking strategies

This document is one of a series addressing Tasks 2 and 3.

### ACKNOWLEDGEMENTS

The University of California Pavement Research Center acknowledges the assistance of the Rubber Pavements Association, Valero Energy Corporation, and Paramount Petroleum which contributed funds and asphalt binders for the construction of the Heavy Vehicle Simulator test track discussed in this study.

The reports prepared during the reflective cracking study document data from construction, Heavy Vehicle Simulator (HVS) tests, laboratory tests, and subsequent analyses. These include a series of first-and second-level analysis reports and two summary reports. On completion of the study this suite of documents will include:

- 1. Reflective Cracking Study: Summary of Construction Activities, Phase 1 HVS testing and Overlay Construction (UCPRC-RR-2005-03).
- Reflective Cracking Study: First-level Report on the HVS Rutting Experiment (UCPRC-RR-2007-06).
- Reflective Cracking Study: First-level Report on HVS Testing on Section 590RF 90 mm MB4-G Overlay (UCPRC-RR-2006-04).
- Reflective Cracking Study: First-level Report on HVS Testing on Section 589RF 45 mm MB4-G Overlay (UCPRC-RR-2006-05).
- Reflective Cracking Study: First-level Report on HVS Testing on Section 587RF 45 mm RAC-G Overlay (UCPRC-RR-2006-06).
- Reflective Cracking Study: First-level Report on HVS Testing on Section 588RF 90 mm AR4000-D Overlay (UCPRC-RR-2006-07).
- Reflective Cracking Study: First-level Report on HVS Testing on Section 586RF 45 mm MB15 Overlay (UCPRC-RR-2006-12).
- Reflective Cracking Study: First-level Report on HVS Testing on Section 591RF 45 mm MAC15TR-G Overlay (UCPRC-RR-2007-04).
- 9. Reflective Cracking Study: HVS Test Section Forensic Report (UCPRC-RR-2007-05).
- Reflective Cracking Study: First-level Report on Laboratory Fatigue Testing (UCPRC-RR-2006-08).
- 11. Reflective Cracking Study: First-level Report on Laboratory Shear Testing (UCPRC-RR-2006-11).
- Reflective Cracking Study: Back Calculation of FWD Data from HVS Test Sections (UCPRC-RR-2007-08).
- 13. Reflective Cracking Study: Second-Level Analysis Report (UCPRC-RR-2007-09).
- 14. Reflective Cracking Study: Summary Report (UCPRC-SR-2007-01). Detailed summary report.
- 15. Reflective Cracking Study: Summary Report (UCPRC-SR-2007-03). Four page summary report.

### **CONVERSION FACTORS**

SI* (MODERN METRIC) CONVERSION FACTORS							
APPROXIMATE CONVERSIONS TO SI UNITS							
Symbol	Convert From	Multiply By	Convert To	Symbol			
LENGTH							
in	inches	inches 25.4 millimeters mn					
ft	feet	0.305	meters	m			
		AREA					
in <sup>2</sup>	square inches	645.2	square millimeters	mm <sup>2</sup>			
ft <sup>2</sup>	square feet	0.093	square meters	m2			
		VOLUME					
ft <sup>3</sup>	cubic feet	0.028	cubic meters	m <sup>3</sup>			
		MASS					
lb	pounds	0.454	kilograms	kg			
	TEMPER	ATURE (exact degree	es)				
°F	Fahrenheit	5 (F-32)/9	Celsius	С			
		or (F-32)/1.8					
	FORCE and	d PRESSURE or STR	ESS				
lbf	poundforce	4.45	newtons	Ν			
lbf/in <sup>2</sup>	poundforce/square inch	6.89	kilopascals	kPa			
	APPROXIMATE C	CONVERSIONS FRO	M SI UNITS				
Symbol         Convert From         Multiply By         Convert To							
	LENGTH						
mm	millimeters	0.039	inches	in			
m	meters	3.28	feet	ft			
		AREA					
mm <sup>2</sup>	square millimeters	0.0016	square inches	in <sup>2</sup>			
m <sup>2</sup>	square meters	10.764	square feet	ft <sup>2</sup>			
		VOLUME					
m <sup>3</sup>	cubic meters	35.314	cubic feet	ft <sup>3</sup>			
	I	MASS					
kg	kilograms	2.202	pounds	lb			
	TEMPER	ATURE (exact degree	es)				
С	Celsius	1.8C+32	Fahrenheit	F			
	FORCE and	d PRESSURE or STR	ESS				
N	newtons	0.225	poundforce	lbf			
kPa	kilopascals	0.145	poundforce/square inch	lbf/in <sup>2</sup>			

\*SI is the symbol for the International System of Units. Appropriate rounding should be made to comply with Section 4 of ASTM E380. (Revised March 2003)

# **GLOSSARY OF TERMS**

av	Percent air-void content
binder	Binder types including AR4000, ARB, MB4, MB15, and MAC15
сотр	Compaction including FMFC, FMLC, and LMLC
cond	Conditioning, either aging or non-aging
grad	Gradation
FMFC	Field-mixed field-compacted
FMLC	Field-mixed laboratory-compacted
LMLC	Laboratory-mixed laboratory-compacted
$G^*$	Resilient shear modulus
$\ln \alpha_1$ and $\beta_1$	Intercept and slope of Stage I of a three-stage fatigue/shear Weibull curve
$\ln \alpha_2$ and $\beta_2$	Intercept and slope of Stage II of a three-stage fatigue/shear Weibull curve
$\ln \alpha_3$ and $\beta_3$	Intercept and slope of Stage III of a three-stage fatigue/shear Weibull curve
lnG	Initial resilient shear modulus (MPa) in natural logarithm
lnkcy5	Permanent shear strain after 5,000 loading cycles
$\ln n_1$	Separation point between Stage I and Stage II of a three-stage fatigue/shear Weibull
	curve
$\ln n_2$	Separation point between Stage II and Stage III of a three-stage fatigue/shear Weibull
	curve
ln <i>Nf</i>	Traditional fatigue life (repetitions at 50 percent loss of initial stiffness) in natural
	logarithm
lnpct5	Cycles to 5 percent permanent shear strain (in natural logarithm)
ln <i>stif</i>	Initial stiffness (MPa) in natural logarithm
lnstn	Strain level in natural logarithm
lnsts	Stress level (kPa) in natural logarithm
ра	Phase angle
PSS	Permanent shear strain
RSS	Residual sum of squares
SR	Stiffness ratio
$srn_1$	Stage I stiffness ratio in a three-stage fatigue Weibull curve
$srn_2$	Stage II stiffness ratio in a three-stage fatigue Weibull curve
temp	Temperature in °C
$\gamma_1$	Parameter that determines the degree of slope change from Stage I to Stage II of a three-
	stage fatigue/shear Weibull curve
$\gamma_2$	Parameter that determines the degree of slope change from Stage II to Stage III of a
	three-stage fatigue/shear Weibull curve

This report is one in a series of first-level analysis reports that describe the results of HVS and laboratory testing on a full-scale experiment being performed at the Richmond Field Station (RFS) to validate Caltrans overlay strategies for the rehabilitation of cracked asphalt concrete. It describes the results of the laboratory shear tests on mixes used as overlays in the experiment. The testing forms part of Partnered Pavement Research Center Strategic Plan Element 4.10: "Development of Improved Rehabilitation Designs for Reflective Cracking."

The objective of this project is to develop improved rehabilitation designs for reflective cracking for California. This objective will be met after completion of the following four tasks:

- 1. Develop improved mechanistic models of reflective cracking in California
- 2. Calibrate and verify these models using laboratory and HVS testing
- 3. Evaluate the most effective strategies for reflective cracking
- 4. Provide recommendations for reflective cracking strategies

This report is one of a series addressing Tasks 2 and 3. It consists of three main chapters. Chapter 2 provides an overview of the experimental design. Chapter 3 summarizes the results on binder tests, conducted by the Federal Highways Administration. Chapter 4 details the shear testing exercise and includes, temperature, air-void content, aging, mixing and compaction, and gradation effects, as well the analysis procedures followed and the results obtained. Comparison of the laboratory and test section performance, including the results of a forensic investigation to be conducted when testing is complete, will be discussed in second-level reports once the data from all of the studies has been collected. It must be emphasized that the study was focused on the use of modified binders in thin overlays on existing cracked asphalt surfaces and not in structural layers. The use of modified binders in thick overlays or as structural layers is currently not recommended.

Five binders were included in this study: AR4000 asphalt cement, asphalt rubber, and three modified binders termed MB4, MB15, and MAC15. The MB4 binder meets the Caltrans MB4 specification, as of 2003. The binder referred to as MB15 also meets the MB4 specification and contains 15 percent recycled rubber. The binder referred to as MAC15 meets the Southern California Greenbook specification (2003) for MAC15TR. The modified binders were blended at the terminal.

All mixes used the same aggregate source, and common aggregate gradations were used for all densegraded mixes and all gap-graded mixes. The AR4000 binder was used in a dense-graded asphalt concrete (DGAC) mix, and the asphalt rubber binder was used in a gap-graded rubber asphalt concrete (RAC-G) mix. In most of the experiments included in this laboratory study the modified binders were used in gapgraded mixes. These mixes were the same as those placed for accelerated pavement testing using the Heavy Vehicle Simulator (HVS). Samples of the five mixes were prepared from loose mix samples obtained at the time of the overlay construction and stored in sealed containers until ready for compaction and testing. These resulting specimens have been designated in the report as field-mixed, laboratorycompacted (FMLC). The majority of the shear test results presented in this report are from FMLC specimens. A comparison was also made between dense-graded mixes with the three modified binders and the RAC-G and DGAC mixes because of the potential for using the modified binders in dense-graded as well as in gap-graded mixes. For this comparison, raw binder and aggregate samples retained since construction were used to mix and compact specimens. These specimens were referred to as laboratorymixed, laboratory-compacted (LMLC).

A comprehensive experimental design was prepared for the study. To test a full factorial considering all the variables over 2,800 tests would have been required. Because of time and fund constraints, a partial factorial experiment was completed with 186 tests.

As-built binder contents of the field mixes were determined from ignition oven tests, after calibration using raw aggregate samples. The design binder contents for the DGAC and RAC-G mixes are based on Caltrans mix design requirements (Section 39 of the Standard Specifications for the DGAC and Section 39-10 of the Standard Special Provisions for the RAC-G). Design binder contents for the gap-graded mixes with the MB4, MB15, and MAC15 binders were selected based on Caltrans mix design requirements. For the LMLC dense-graded mixes containing the modified binders, the standard California procedure for mix design was followed to define the binder contents used for the test specimens.

Binder tests were performed for four of the binders (AR4000, MB4, MB15, and MAC15) by the Federal Highway Administration using the Bending Beam Rheometer (BBR) and the Dynamic Shear Rheometer (DSR) over a range of loading times for the BBR and frequencies for the DSR. Specimens were tested in their original condition, after short-term aging using the Rolling Thin Film Oven (RTFO) Test, and after long-term aging using the Pressure Aging Vessel (PAV) Test. Based on the current specification requirements, binder rankings considering low-temperature cracking, fatigue, and rutting are as follows, ranked from least to highest susceptibility:

Low Temperature Cracking	Fatigue	Permanent Deformation
MB4	MB4	AR4000
MB15	MB15	MB4 and MB15
MAC15	MAC15	MAC15
AR4000	AR4000	

Significant factors affecting shear response have been identified using:

- A correlation matrix,
- Analysis of variance (ANOVA),
- Design plots, and
- Pairs diagrams.

This approach was deemed essential since a partial factor experiment (186 tests) rather than a full factorial (2,880 tests) was conducted. By using this approach, there is greater confidence that the major effects are included in any performance equation resulting from the experiment to predict the performance of a mix containing a specific binder in pavement structures subjected to different traffic and climate conditions.

Regression models are presented for Cycles to 5 Percent Permanent Shear Strain (PSS), Permanent Shear Strain (PSS) at 5,000 Cycles, and resilient shear modulus ( $G^*$ ) for the various mixes tested. Results predicted by the regression equations are presented for different values of the input variables (stress, temperature, etc.). It must be emphasized that when these regression equations are used for pavement performance analyses, mixes similar to those used in this investigation and within the range of the variables used to calibrate the equations should be used in order to obtain reasonable estimates of the effects of the various binders on pavement performance.

Based on the shear test results for FMLC specimens from the mixes used in the overlay experiment, mix rankings for Cycles to 5 percent PSS, PSS at 5,000 Cycles, and  $G^*$  are as follows, from best expected rutting performance to worst:

Cycles to 5	<b>Resilient Shear</b>
Percent PSS	Modulus $(G^*)$
AR4000-D	AR4000-D
MAC15-G	RAC-G
RAC-G	MAC15-G
MB4-G	MB4-G
MB15-G	MB15-G
	Cycles to 5 Percent PSS AR4000-D MAC15-G RAC-G MB4-G MB15-G

While the shear tests on the laboratory mixed, laboratory-compacted dense-graded mixes containing the three modified binders were limited, the performance of these three dense-graded mixes was generally better than those of the corresponding gap-graded mixes. The results for the dense-graded AR4000 mix in the gradation study were not consistent relative to those of the RAC-G mix. The difference may be due to differences in aging between FMLC and LMLC specimens resulting from reheating for compaction of the field-mix for the FMLC specimens.

In conclusion, it must be emphasized that until a range of pavement types and environments are evaluated in the second-level analysis, only a general indication of the expected relative rutting performance of the modified binders can be deduced. It would appear that the MB4 and MB15 binders used in gap-graded mixes have a somewhat greater risk of rutting at high temperatures compared to RAC-G mixes, while gap-graded mixes with MAC15 binder had results similar to those of RAC-G. Recommendations for the use of MB4 and MB15 materials in thicker layers and as dense-graded mixes await further test results and pavement performance analyses.

# TABLE OF CONTENTS

GLO	SSARY	OF TEI	RMS	v
EXE	CUTIV	E SUMN	1ARY	vii
LIST	OF TA	BLES		xiii
LIST	OF FIG	GURES.		XV
1.	INTE	RODUC	ΤΙΟΝ	1
	1.1.	Object	tives	1
	1.2.	Overal	Il Project Organization and Deliverables	1
	1.3.	Conter	nt and Structure of this Report	4
	1.4.	Measu	rement Units	4
2.	EXPI	ERIMEN	NT DESIGN	5
	2.1.	Introd	uction	5
	2.2.	Test P	rocedure	5
	2.3.	Experi	iment Design	6
		2.3.1	Temperature and Shear Stress Effects (FMLC)	10
		2.3.2	Air-Void Content Effect (FMLC)	10
		2.3.3	Aging Effect (FMLC)	10
		2.3.4	Mixing and Compaction Effect (FMLC, LMLC, and FMFC)	10
		2.3.5	Gradation Effect (LMLC)	11
	2.4.	Specir	nen Preparation	11
		2.4.1	Laboratory-Mixed, Laboratory-Compacted Specimens	11
		2.4.2	Field-Mixed, Laboratory Compacted Specimens	14
		2.4.3	Field-Mixed, Field-Compacted Specimens	14
	2.5.	Ignitio	on Oven Tests	14
		2.5.1	Test Method	14
		2.5.2	Results	15
3.	BINI	DER TES	STING	17
	3.1.	Introd	uction	17
	3.2.	Dynan	nic Shear Rheometer	17
		3.2.1	Test Method	17
		3.2.2	Results	17
		3.2.3	Master Curves of Shear Complex Modulus	
	3.3.	Bendi	ng Beam Rheometer	25
		3.3.1	Test Method	25
		3.3.2	Results	25

4.	SHEA	AR TES	STING	
	4.1.	Introd	luction	
		4.1.1	Definitions Used in Statistical Analyses	
		4.1.2	Expected Effects of Response Variables on Performance	
		4.1.3	Presentation of Results	
	4.2.	Tempe	erature Effect	
		4.2.1	Results	
		4.2.2	Identification of Significant Factors	
		4.2.3	Regression Analysis	
	4.3.	Air-V	oid Content Effect	41
	4.4.	Aging	g Effect	
	4.5.	Mixin	g and Compaction Effect	
	4.6.	Grada	tion Effect	51
	4.7.	Pooled	d Shear Tests	
	4.8.	Summ	nary of Factor Identification	
	4.9.	Summ	nary of Regression Analysis	60
		4.9.1	Cycles to 5 Percent Permanent Shear Strain	
		4.9.2	PSS at 5,000 Cycles	
		4.9.3	Resilient Shear Modulus $(G^*)$	
	4.10.	Weibu	ull Analysis	
	4.11.	Secon	d-Level Analysis	69
5.	CON	CLUSI	ONS	71
	5.1.	Findin	ngs and Observations	71
	5.2.	Recon	nmendations	73
6.	REFF	ERENC	ES	
APPE	NDIX A	A: SUM	IMARY OF RESULTS	77
APPE	NDIX H	B: MIX	DESIGN INFORMATION	
APPE	NDIX (	C: UCP	RC MB-DENSE MIX DESIGN SUMMARY	

# LIST OF TABLES

Table 2.1:	Overall Laboratory Shear Testing Test Plan	7
Table 2.2:	Experimental Design for Laboratory Shear Testing	8
Table 2.3:	Summary of Gradation Curves	11
Table 2.4:	Design Binder Contents of Laboratory Mixes	12
Table 2.5:	LMLC Binder Mixing Temperatures	13
Table 2.6:	Compaction Temperatures for LMLC and FMLC	14
Table 2.7:	Summary of Asphalt Contents from Binder Ignition Tests	15
Table 3.1:	Summary of SSV and SSD Values from DSR Test Results	20
Table 3.2:	Summary of Bending Beam Rheometer Test Results	25
Table 4.1:	Summary of Expected Effects of Response Variables from RSST on Performance	30
Table 4.2:	Correlation Matrix for Temperature Effect	34
Table 4.3:	Analysis of Variance for Temperature Effect	34
Table 4.4:	Contrast Tables of Category Covariates Used in Regression Analyses	39
Table 4.5:	Regression Models for In Cycles to 5 Percent Permanent Shear Strain	60
Table 4.6:	Regression Models for In Permanent Shear Strain at 5,000 Cycles	61
Table 4.7:	Regression Models for Resilient Shear Modulus (G*)	61
Table A.1:	Summary of Shear Laboratory Test Results for AR4000 Mixes (Temperature Effect)	78
Table A.2:	Summary of Shear Laboratory Test Results for RAC Mixes (Temperature Effect)	79
Table A.3:	Summary of Shear Laboratory Test Results for MAC15 Mixes (Temperature Effect)	80
Table A.4:	Summary of Shear Laboratory Test Results for MB15 Mixes (Temperature Effect)	81
Table A.5:	Summary of Shear Laboratory Test Results for MB4 Mixes (Temperature Effect)	82
Table A.6:	Summary of Shear Laboratory Test Results for all Mixes (Air-Void Content Effect)	83
Table A.7:	Summary of Shear Laboratory Test Results for all Mixes (Aging Effect).	84
Table A.8:	Summary of Shear Laboratory Test Results for all Mixes (Compaction Effect)	85
Table A.9:	Summary of Shear Laboratory Test Results for all Mixes (Gradation Effect)	86
Table C.1:	MB-15 Stability Values	94
Table C.2:	MB-15 Air Voids Analysis	95
Table C.3:	MB-4 Stability Values	95
Table C.4:	MB-4 Air Voids Analysis	96
Table C.5:	MAC-15 Stability Values	96
Table C.6:	MAC-15 Air Voids Analysis	97
Table C.7:	OBC Summary	97

# LIST OF FIGURES

Figure 1.1: Timeline for the Reflective Cracking Study	3
Figure 2.1: Gradation curves for gap-graded mixes	12
Figure 2.2: Gradation curves for dense-graded mixes.	12
Figure 3.1: G/sinð summary of DSR test results on original binder	18
Figure 3.2: G/sinð summary of DSR test results on RTFO aged binder	18
Figure 3.3: Gsino summary of DSR test results on PAV-aged binder	19
Figure 3.4: Master curves of shear complex modulus of AR4000.	22
Figure 3.5: Temperature-shift relationships of AR4000	22
Figure 3.6: Master curves of shear complex modulus of MB4	22
Figure 3.7: Temperature-shift relationships of MB4.	22
Figure 3.8: Master curves of shear complex modulus of MB15	23
Figure 3.9: Temperature-shift relationships of MB15.	23
Figure 3.10: Master curves of shear complex modulus of MAC15	23
Figure 3.11: Temperature-shift relationships of MAC15.	23
Figure 3.12: Comparison of G* Master curves (original)	24
Figure 3.13: Comparison of G* Master curves (RTFO).	24
Figure 3.14: Comparison of G* Master curves (PAV).	24
Figure 3.15: Creep stiffness summary of BBR test results	
Figure 3.16: Summary of m-value results from BBR tests	
Figure 4.1: Summary plots of temperature effect and Cycles to 5 Percent PSS	31
Figure 4.2: Summary plots of temperature effect and PSS at 5,000 Cycles	
Figure 4.3: Summary plots of temperature effect and Resilient Shear Modulus, G*	
Figure 4.4: Design plots for temperature effect (6 percent AV).	35
Figure 4.5: Summary statistics of main effects	
Figure 4.6: Summary boxplots of air-void content effect and Cycles to 5 Percent PSS	
Figure 4.7: Summary boxplots of air-void content and PSS at 5,000 Cycles	
Figure 4.8: Summary boxplots of air-void content and Resilient Shear Modulus (G*).	43
Figure 4.9: Design plots of air-void content effect.	43
Figure 4.10: Summary boxplots of aging effect and Cycles to 5 Percent PSS.	45
Figure 4.11: Summary boxplots aging effect and PSS at 5,000 Cycles	45
Figure 4.12: Summary boxplots of aging effect and Resilient Shear Modulus (G*.)	46
Figure 4.13: Design plots for aging effect.	46

Figure 4.14:	Summary boxplots of compaction effect and Cycles to 5 Percent PSS	48
Figure 4.15:	Summary boxplots of compaction effect and PSS at 5,000 Cycles	49
Figure 4.16:	Summary boxplots of compaction effect and Resilient Shear Modulus (G*).	49
Figure 4.17:	Design plots for compaction effect	50
Figure 4.18:	Summary boxplots of gradation effect and Cycles to 5 Percent PSS	52
Figure 4.19:	Summary boxplots of gradation effect and PSS at 5,000 Cycles	52
Figure 4.20:	Summary boxplots of gradation effect and Resilient Shear Modulus (G*)	53
Figure 4.21:	Design plots for gradation effect.	53
Figure 4.22:	Pairs diagram showing relationships among variables	55
Figure 4.23:	Summary boxplots of pooled shear testing for permanent shear strain at cycles	56
Figure 4.24:	Summary boxplots of pooled shear testing for Cycles to 5 Percent PSS	56
Figure 4.25:	Example design plots for pooled shear tests	57
Figure 4.26:	Residual plots of Ln(pct5). (Pooled Shear Tests)	58
Figure 4.27:	Residual plots of Ln(kcyc5) (Pooled Shear Tests).	59
Figure 4.28:	Residual plots of LnG. (Pooled Shear Tests).	59
Figure 4.29:	Schematic summary of Cycles to 5 Percent PSS regression models - Part A	63
Figure 4.30:	Schematic summary of Cycles to 5 Percent PSS regression models - Part B	64
Figure 4.31:	Schematic summary of PSS at 5,000 Cycles regression models - Part A	64
Figure 4.32:	Schematic summary of PSS at 5,000 Cycles regression models - Part B	65
Figure 4.33:	Schematic summary of Resilient Shear Modulus (G*) regression models - Part A	65
Figure 4.34:	Schematic summary of Resilient Shear Modulus (G*) regression models - Part B	66
Figure 4.35:	Two-Stage Weibull fitting for RSST-CH test result.	67
Figure 4.36:	PSS Correlation.	68
Figure 4.37:	Correlation between PSS at 5,000 Cycles and Cycles to 5 Percent PSS	69
Figure B.1:	Caltrans DGAC mix design	87
Figure B.2:	Caltrans RAC-G mix design	88
Figure B.3:	Caltrans MB-4 mix design	89
Figure B.4:	Caltrans MB-15 mix design	90
Figure B.5:	Caltrans MAC 15 mix design.	91
Figure C.1:	Stability vs. Percent Asphalt Content	98
Figure C.2:	Air Voids vs. Percent Asphalt Content	98

### **1. INTRODUCTION**

#### 1.1. Objectives

The first-level analysis presented in this report is part of Partnered Pavement Research Center Strategic Plan Element 4.10 (PPRC SPE 4.10) being undertaken for the California Department of Transportation (Caltrans) by the University of California Pavement Research Center (UCPRC). The objective of the study is to evaluate the reflective cracking performance of asphalt binder mixes used in overlays for rehabilitating cracked asphalt concrete pavements in California. The study includes mixes modified with rubber and polymers, and it will develop tests, analysis methods, and design procedures for mitigating reflective cracking in overlays. This work is part of a larger study on modified binder (MB) mixes being carried out under the guidance of the Caltrans Pavement Standards Team (PST) (1), which includes laboratory and accelerated pavement testing using the Heavy Vehicle Simulator (carried out by the UCPRC), and the construction and monitoring of field test sections (carried out by Caltrans).

#### 1.2. Overall Project Organization and Deliverables

This UCPRC project is a comprehensive study, carried out in three phases, involving the following primary elements (2):

- Phase 1
  - The construction of a test pavement and subsequent overlays;
  - Six separate Heavy Vehicle Simulator (HVS) tests to crack the pavement structure;
  - Placing of six different overlays on the cracked pavement;
- Phase 2
  - Six HVS tests to assess the susceptibility of the overlays to high-temperature rutting (Phase 2a);
  - Six HVS tests to determine the low-temperature reflective cracking performance of the overlays (Phase 2b);
  - Laboratory shear and fatigue testing of the various hot-mix asphalts (Phase 2c);
  - Falling Weight Deflectometer (FWD) testing of the test pavement before and after construction and before and after each HVS test;
  - Forensic evaluation of each HVS test section;
- Phase 3
  - Performance modeling and simulation of the various mixes using models calibrated with data from the primary elements listed above.

#### Phase 1

In this phase, a conventional dense-graded asphalt concrete (DGAC) test pavement was constructed at the Richmond Field Station (RFS) in the summer of 2001. The pavement was divided into six cells, and within each cell a section of the pavement was trafficked with the HVS until the pavement failed by either fatigue  $(2.5 \text{ m/m}^2 [0.76 \text{ ft/ft}^2])$  or rutting (12.5 mm [0.5 in]). This period of testing began in the fall of 2002 and was concluded in the spring of 2003. In June 2003 each test cell was overlaid with either conventional DGAC or asphalt concrete with modified binders as follows:

- Full-thickness (90 mm) AR4000-D overlay, included as a control for performance comparison purposes;
- Full-thickness (90 mm) MB4 gap-graded overlay;
- Half-thickness (45 mm) rubberized asphalt concrete gap-graded overlay (RAC-G), included as a control for performance comparison purposes;
- Half-thickness (45 mm) MB4 gap-graded overlay;
- Half-thickness (45 mm) MB4 gap-graded overlay with minimum 15 percent recycled tire rubber, and
- Half-thickness (45 mm) MAC15TR gap-graded overlay with minimum 15 percent recycled tire rubber.

The conventional overlay was designed using the current (2003) Caltrans overlay design process. The various modified overlays were either full (90 mm) or half thickness (45 mm). Mixes were designed by Caltrans. The overlays were constructed in one day.

#### Phase 2

Phase 2 included high-temperature rutting and low-temperature fatigue testing with the HVS as well as laboratory shear and fatigue testing. The rutting tests were started and completed in the fall of 2003. For these tests, the HVS was placed above a section of the underlying pavement that had not been trafficked during Phase 1. A low-temperature fatigue test was next conducted on each overlay from the winter of 2003-2004 to the summer of 2007. For these tests, the HVS was positioned precisely on top of the sections of failed pavement from the Phase 1 HVS tests to investigate the extent and rate of crack propagation through the overlay.

In conjunction with Phase 2 HVS testing, a full suite of laboratory testing, including shear and fatigue testing, was carried out on field-mixed/field-compacted, field-mixed/laboratory-compacted, and laboratory-mixed/laboratory-compacted specimens.

#### Phase 3

Phase 3 entailed a second-level analysis carried out on completion of HVS and laboratory testing. This included extensive analysis and characterization of the mix fatigue and mix shear data, backcalculation of the FWD data, performance modeling of each HVS test, and a detailed series of pavement simulations carried out using the combined data.

An overview of the project timeline is shown in Figure 1.1.



Figure 1.1: Timeline for the Reflective Cracking Study.

The reports prepared during the reflective cracking study document data from construction, HVS tests, laboratory tests, and subsequent analyses. These include a series of first- and second-level analysis reports and two summary reports. On completion of the study this suite of documents will include:

- One first-level report covering the initial pavement construction, the six initial HVS tests, and the overlay construction (Phase 1);
- One first-level report covering the six Phase 2 rutting tests. This report offers no detailed explanations or conclusions on the performance of the pavements;
- Six first-level reports, each covering a single Phase 2 fatigue test. These reports contain summaries and trends of the measured environmental conditions, pavement responses, and pavement performance. They offer no detailed explanations or conclusions on the performance of the pavement.
- One first-level report covering laboratory shear testing;
- One first-level report covering laboratory fatigue testing;
- One report summarizing the HVS test section forensic investigation;
- One report detailing Falling Weight Deflectometer (FWD) results and analysis;

- One second-level analysis report detailing characterization of laboratory fatigue and shear data, pavement modeling analysis, comparisons of the various overlays, and simulations using various scenarios (Phase 3), and
- A four-page summary report capturing the conclusions of the experiment and a longer, more detailed summary report that covers the findings and conclusions from the research conducted by the UCPRC.

Reports are prepared as soon as a specific HVS or laboratory test is complete. Additional findings from forensic investigations and later analysis are covered in the forensic, second-level analysis, and summary reports.

#### **1.3.** Content and Structure of this Report

This report presents a summary of the results of the laboratory shear test program, detailed results of which are available in the UCPRC relational database. The report is organized as follows:

- Chapter 2 details the test plan and describes specimen preparation and conditioning.
- Chapter 3 provides information on the binders used in the study.
- Chapter 4 presents and discusses the results of shear testing in terms of the variables listed above.
- Chapter 5 provides conclusions.
- Appendix A contains summary tables of test results.
- Appendix B contains mix design summary data.

#### **1.4.** Measurement Units

Metric units have always been used in the design and layout of HVS test tracks, all the measurements and data storage, and all associated laboratory testing at the eight HVS facilities worldwide (as well as all other international accelerated pavement testing facilities). Use of the metric system facilitates consistency in analysis, reporting, and data sharing.

In this report, metric and English units (provided in parentheses after the metric units) are used in the Executive Summary, Chapter 1 and 2, and the Conclusion. In keeping with convention, only metric units are used in Chapters 3, 4, and 5. A conversion table is provided on Page iv at the beginning of this report.

### 2. EXPERIMENT DESIGN

#### 2.1. Introduction

The laboratory program included shear tests on mixes used in the following accelerated pavement testing sections:

- 1. Full-thickness (90 mm [3.5 in]) dense-graded asphalt concrete (DGAC) with AR-4000 binder, included as a control for performance comparison purposes
- 2. Half-thickness (45 mm [1.7 in]) rubberized asphalt concrete gap-graded (RAC-G) overlay, included as a control for performance comparison purposes
- 3. Full-thickness (90 mm) MB4 gap-graded overlay
- 4. Half-thickness (45 mm) MB4 gap-graded overlay
- 5. Half-thickness (45 mm) MB4 gap-graded overlay with minimum 15 percent recycled tire rubber (referred to as "MB15" in this report)
- 6 Half-thickness (45 mm) MAC15TR gap-graded overlay with minimum 15 percent recycled tire rubber (referred to as "MAC15" in this report)

Samples of loose asphalt mix were collected from the HVS test site during construction of the test sections. Specimens compacted in the laboratory using this material are referred to as field-mixed, laboratory-compacted (FMLC) specimens. Samples of the asphalt binders and aggregates were obtained at the hot-mix plant during construction. These materials were used to prepare laboratory-mixed, laboratory-compacted (LMLC) specimens. Cores were also cut from the pavement section for testing, and are referred to as field-mixed, field-compacted (FMFC) specimens. The resulting specimens were used to evaluate the influence of binder type, applied shear stress, temperature, degree of compaction (air-void content), aging, and aggregate gradation on permanent deformation performance.

Summaries of the test procedures, experiment design, and specimen preparation are included in this chapter.

#### 2.2. Test Procedure

The laboratory shear test used in this study was AASHTO T320, "Determining the Permanent Shear Strain and Stiffness of Asphalt Mixtures Using the Superpave Shear Tester (SST), Procedure C, Repeated

Shear Test at Constant Height (RSST-CH)." This test procedure was originally developed as part of the Strategic Highway Research Program (SHRP). An ASTM version of the test has recently been approved.

In this report, the test is referred to as the RSST (Repeated Simple Shear Test).

In the standard test methodology, cylindrical test specimens 50 mm (2 in) thick by 150 mm (6 in) in diameter are subjected to repeated loading in shear using a 0.1-second haversine waveform followed by a 0.6-second rest period. A shear stress equaling 70 kPa (10 psi) is applied while the permanent (unrecoverable) and recoverable shear strains are measured. The permanent shear strain versus applied repetitions is normally recorded up to a value of 5 percent although 5,000 repetitions are called for in the AASHTO procedure. A constant temperature is maintained during the test (termed the *critical temperature*) representative of the local environment, generally in the temperature range of 40°C to 60°C (104°F to 140°F).

For this study, the test was also performed at two stresses greater than that used in the standard test and at two temperatures. Since tests run to 5,000 repetitions (as in the AASHTO procedure) may not produce significant permanent shear deformation, tests were run up to 30,000 repetitions or until 5 percent permanent shear strain was achieved. When specimens did not reach 5 percent permanent shear strain within 30,000 repetitions, results were extrapolated to this strain level. The purpose of the additional stress levels and temperatures, and the extended tests, was to obtain more a comprehensive data set for later analysis and simulation of the permanent deformation performance of the mixes.

#### 2.3. Experiment Design

The experiment design was formulated to quantify the effects of:

- Applied shear stress,
- Temperature,
- Degree of compaction (air voids),
- Mix aging,
- Mixing and compaction method, and
- Aggregate gradation.

Table 2.1 presents an overview of the shear experiment design. Table 2.2 provides the detailed experiment designs for the study. The following sections briefly discuss the effects mentioned, and the

motivation for and application of the study. With each effect, the type of specimen tested (laboratorymixed, laboratory-compacted [LMLC]; field-mixed, laboratory-compacted [FMLC]; or field-mixed, fieldcompacted [FMFC]) is noted in parentheses. LMLC specimens were prepared from aggregate and asphalt samples taken at the plant and refinery during construction, and later mixed and compacted in the laboratory. FMLC specimens were compacted in the laboratory using mix collected from the plant during construction of the HVS test section overlays. FMFC specimens, field cores, were obtained from cores extracted from the pavement section after construction of the HVS test section overlays.

Mix/ Compaction	AV*	AC**	Grad.	Test Type	Variables	Total Tests		
FMLC	Design AV (6±0.5%)	Field AC	Gap- graded /Dense- graded	RSST	2 temperatures (45, 55°C) 3 stress levels (70, 100, 130 kPa) 3 replicates	18		
(Temperature Susceptibility)	Field AV (9±1%)	Field AC	Gap- graded /Dense- graded	RSST	2 temperature (45, 55°C) 1 stress levels (70 kPa) 2 replicates	4		
FMLC	Design AV	Field	Gap- graded	RSST	2 temperature (45,55°C) LTOA (6 days) 1 stress level (70 kPa) 2 replicates	4		
(Aging)	(6±0.5%)	(6±0.5%)	(6±0.5%)	AC	AC graded	graded	1 temperature (45°C) LTOA (6 days) 2 stress levels (100, 130 kPa) 2 replicates	4
LMLC	Design AV (6±0.5%)	Design AC	Gap- graded	RSST	1 temperature (45°C) 1 stress level (70 kPa) 2 replicates	2		
LMLC	Design AV (6±0.5%)	Design AC (TBD)	Dense- graded	RSST	1 temperature (45°C) 1 strain levels (70 kPa) 2 replicates	2		
Total tests per mix type						39 (45)		
					5 mixes	197* (227)		
*AV ·	- Air void	*:	*AC - Aspha	alt content	(percent by mass of aggregate)			

 Table 2.1: Overall Laboratory Shear Testing Test Plan

In order to test a full factorial, a total of 2,880 tests (three replicates of five binder types, two mix types, two compaction types, two condition types, two gradations, two air-void contents, two temperatures, and three stress levels) would be required. This number of tests was unrealistic in terms of both time and resources. A partial factorial was therefore tested and where possible the same tests to evaluate different effects were not repeated. As noted, results were extrapolated when required.

Type of Shear Study	Test Type	Compaction	Condition	Binder Type	Gradation	Design Asphalt Content (%)*	Air-Void Content (%)	Temp. (°C)	Stress (kPa)	Replicates	Total Tests
				AR4000	DG	5.0					2 x 3 x 3 = 18
Temperature				RAC		8.0					2 x 3 x 3 = 18
Effect	RSST-CH	FMLC	none	MAC15		7.4	$6 \pm 0.5$	45, 55	70,100, 130	3	2 x 3 x 3 = 18
(90)				MB15	00	7.1					2 x 3 x 3 = 18
				MB4		7.2					2 x 3 x 3 = 18
Air-Void				AR4000	DG	5.0					$2 \ge 1 \ge 2 = 4$
Content Effect			RAC		8.0					$2 \ge 1 \ge 2 = 4$	
(20)	RSST-CH	FMLC	none	MAC15		7.4	9 ± 1	45, 55	70	2	$2 \ge 1 \ge 2 = 4$
(Compared to Temp Effect				MB15	GG	7.1					$2 \ge 1 \ge 2 = 4$
at 6%)				MB4		7.2					$2 \ge 1 \ge 2 = 4$
				AR4000	DG	5.0					2 x 1 x 2 = 4
(20)				RAC		8.0		$6 \pm 0.5$ 45, 55 $70,100, \\ 130$ 3 $9 \pm 1$ 45, 55       70       2 $6 \pm 0.5$ 45, 55       70       2		$2 \ge 1 \ge 2 = 4$	
(Compared to	RSST-CH	FMLC	aging	MAC15	- GG	7.4	$6 \pm 0.5$		70	2	$2 \ge 1 \ge 2 = 4$
Specimens				MB15		7.1					$2 \ge 1 \ge 2 = 4$
at 0%)				MB4		7.2					$2 \ge 1 \ge 2 = 4$
* Design Asphal	t Contents are p	bercent by mass of	aggregate						-		

 Table 2.2: Experimental Design for Laboratory Shear Testing

Type of Shear Study	Test Type	Compaction	Condition	Binder Type	Gradation	Design Asphalt Content (%)	Air-Void Content (%)	Temp. (°C)	Stress (kPa)	Replicates	Total Runs
Compaction Effect		FMFC	Field Aged and	MB4	GG	7.2	$6 \pm 0.5$	45	70	3	$1 \ge 1 \ge 3 = 3$
(26)			Trafficked	AR4000	DG	5.0					$1 \ge 1 \ge 3 = 3$
(Compared to	RSST-CH			AR4000	DG	5.0		= 0.5 45	70	3	$1 \ge 1 \ge 3 = 3$
Temp Effect		LMLC	none M	MB4		7.2	$6 \pm 0.5$				$1 \ge 1 \ge 3 = 3$
Specimens				RAC		8.0					$1 \ge 1 \ge 3 = 3$
at 6%)				MAC15	GG	7.4					$1 \times 1 \times 3 = 3$
ut 0 /0 /				MB15		7.1					$1 \ge 1 \ge 3 = 3$
Gradation Effect (24)				MB4		7.2					$1 \ge 1 \ge 3 = 3$
(Compared to compaction	RSST-CH	LMLC	none	MB15	DG	7.1	$6 \pm 0.5$	45	70	3	$1 \times 1 \times 3 = 3$
effect LMLC specimens)				MAC15		7.4					$1 \times 1 \times 3 = 3$

 Table 2.2: Experimental Design for Laboratory Shear Testing (cont.)

#### **2.3.1** Temperature and Shear Stress Effects (FMLC)

This part of the experiment evaluated the temperature and stress effects on field-mixed, laboratorycompacted (FMLC) specimens. Three replicates at two temperatures (45°C and 55°C [113°F and 131°F]) and three stress levels (70 kPa, 100 kPa and 130 kPa [10 psi, 14.5 psi and 18.8 psi]) were used.

#### 2.3.2 Air-Void Content Effect (FMLC)

The effect of construction quality in terms of compaction on pavement performance was considered by conducting tests on specimens at two different air-void contents,  $6.0 \pm 0.5$  percent and  $9.0 \pm 1.0$  percent. Three replicates at two temperatures (45 C and 55°C) and one stress level (70 kPa) were tested.

#### 2.3.3 Aging Effect (FMLC)

The aging effect simulates extended environmental exposure, generally resulting in stiffening of the binder. For conventional asphalt binders (unmodified), rutting resistance is generally increased as the binder stiffness is increased. In the AASHTO PP2-94 mix aging test, a compacted specimen is conditioned for five days at 85°C (185°F). This period is considered to represent longer term aging in the field. Results from the SHRP program suggest that long-term oven aging at 85°C in a forced draft oven for eight days represents (conservatively) approximate aging at sites nine years or older in the dry-freeze zone, and eighteen years or older in the wet no-freeze zone (4). For this experiment, the aging period was modified to six days at 85°C, based on previous experience (5). After six days of aging in the forced-draft oven, specimens were allowed to cool to room temperature, then conditioned at the shear test temperature for two hours prior to testing.

To evaluate the aging effect of the asphalt binder on permanent deformation, the experiment compared four aged specimens (two temperatures, two replicates) with six non-aged specimens (two temperatures, three replicates), with all of the specimens tested at one air-void content (6 percent) and one test stress (70 kPa). Tests on field cores (field aged and trafficked) were run at a single temperature (45°C) and stress (70 kPa).

#### 2.3.4 Mixing and Compaction Effect (FMLC, LMLC, and FMFC)

In this series the performance of LMLC cores (three replicates of five binder types), FMLC cores (three replicates of five binder types) and FMFC cores (three replicates of two binder types) were compared. One air-void content (6 percent), one applied shear stress (70 kPa), and one test temperature (45°C) were used.

#### 2.3.5 Gradation Effect (LMLC)

HVS testing was conducted only on gap-graded mixes containing the MAC15, MB15 and MB4 binders. The laboratory shear test program was intended to provide information for assessing the use of these modified binders in dense-graded mixes. Mix designs were performed by UCPRC staff according to the CTM 304, 366, and 367 procedures for dense-graded mixes containing the MB4, MB15, and MAC15 binders. Performance of these mixes was compared with that of the dense-graded mix containing the AR4000 binder (DGAC) and the gap-graded mix with the rubber asphalt binder (ARB). One air-void content (6 percent), one applied shear stress (70 kPa), and one test temperature (45°C) were used.

#### 2.4. Specimen Preparation

#### 2.4.1 Laboratory-Mixed, Laboratory-Compacted Specimens

#### Gradation and Binder Contents

Laboratory mix aggregate gradations and binder contents are shown in Table 2.3 and Table 2.4, respectively, and in Figure 2.1 and Figure 2.2. The aggregate gradations conform to the requirements specified by Caltrans Standard Specifications, Section 39. The dense gradation meets standard specification for 19 mm (0.75 in) Type A coarse asphalt concrete (as of 2003) and the gap gradation conforms to the special provisions for Type G-MB. The target dense gradation was determined from field samples for the AR 4000 DGAC mix collected and tested by Caltrans during overlay construction. For the gap gradation, several different field samples from different mixes were tested and the average gradation was calculated and set as the laboratory target.

Siovo Sizo	Gap-	Graded (% pas	sing)	Dense-Graded (% passing)			
(mm)	Design (Caltrans)	Field (Caltrans)	LMLC (UCB)	Design (Caltrans)	Field (Caltrans)	LMLC (UCB)	
25.4	100.0	100.0	100.0	100.0	100.0	100.0	
19.0	98.0	96.3	96.7	98.0	93.0	91.5	
12.7	82.0	78.2	78.2	85.0	72.0	72.4	
9.5	69.0	64.8	64.6	79.0	63.0	63.8	
4.75	36.0	32.3	32.5	49.0	44.0	41.4	
2.38	21.0	20.5	20.2	35.0	31.5	28.8	
1.19	13.0	15.7	15.8	23.0	24.0	23.7	
0.59	10.0	12.3	12.6	16.0	19.0	19.2	
0.23	7.0	9.2	9.2	11.0	13.0	13.4	
0.15	5.0	5.0	5.5	6.0	6.0	6.4	
0.075	3.1	3.6	3.8	4.0	3.7	4.3	

**Table 2.3: Summary of Gradation Curves** 

0		U							
d <sup>1</sup>	Dense-Graded <sup>2</sup>								
nder Content <sup>3</sup>	Binder	Binder Content <sup>3</sup>							
8.0	AR4000	5.0							
7.4	MAC15	6.3							
7.1	MB15	6.2							
7.2	MB4	6.4							
1. Gap-graded mix designs determined by Caltrans.									
2. Dense-graded mix designs for MAC15, MB15, and MB4 binders determined by									
enter, mix design f	or DGAC determined by	v Caltrans.							
	d <sup>1</sup> nder Content <sup>3</sup> 8.0 7.4 7.1 7.2 etermined by Calti gns for MAC15, enter, mix design f	d1Dense-(nder Content <sup>3</sup> Binder8.0AR40007.4MAC157.1MB157.2MB4etermined by Caltrans.gns for MAC15, MB15, and MB4 bienter, mix design for DGAC determined by							

Table 2.4: Design Binder Contents of Laboratory Mixes

3. Percent by mass of aggregate



Figure 2.1: Gradation curves for gap-graded mixes.



Figure 2.2: Gradation curves for dense-graded mixes.

#### Specimen Preparation

Specimens were prepared from raw materials supplied by the contractor constructing the Test Track, Syar Industries, Inc. The aggregate, a basalt, was obtained from Syar's Lake Herman quarry, located near Vallejo, CA. The aggregate blend was obtained from four bins with size ranges as follows: 19 mm x 12.5 mm, 12.5 mm x 9.5 mm, 9.5 mm x dust, and 4.75 mm x dust. Binders used for the test track were obtained from two California refineries, Paramount Petroleum (MAC15) and Valero (all other binders).

The production of shear cores involved:

- Checking the aggregate gradings using AASHTO T11 (wet sieving, passing the 0.075 mm [No 200] sieve) and AASHTO T27 (dry sieving of fine and coarse aggregate)
- Batching of aggregates and mixing with binder
- Short-term oven aging (AASHTO PP2-94)
- Specific gravity testing (AASHTO T209)
- Rolling wheel compaction to produce slabs (PP3-94)
- Coring and sawing to size of cores for shear tests
- Measuring air-void content (AASHTO T275 [Caltrans CTM 308])

In the batching and mixing processes, 7 kg (15.4 lb) batches were heated to the binder-specific mixing temperature (Table 2.5) for at least two hours before mixing. Asphalt binder was heated to the same temperature for approximately one hour, or until it was consistently pourable, then mixed with the aggregate until the aggregates were fully coated (typically in about five minutes). Mixing bowl and blades were preheated to prevent adhesion of the binder.

Mix	<b>Binder Mix Temperatures</b>	Temperature Specification Range					
IVIIX	(° <b>C</b> )	(° <b>C</b> )					
MB4	163 <sup>1</sup>	$150-163^3$					
MB15	163	150–163					
MAC15	163	150–163					
RAC	163	150–163					
AR4000	$145^{2}$	_					
Note: Temperatures for MB mixes are from Caltrans Type D-MB specification.							
<sup>1</sup> 325°F	<sup>2</sup> 295°F <sup>3</sup> 302-325	PF					

**Table 2.5: LMLC Binder Mixing Temperatures** 

The short-term oven aging procedure used in this investigation (AASHTO PP2-94) attempts to replicate aging that occurs in the mixing and compaction process, and perhaps some early in-situ aging. In this procedure, oven-aging involves conditioning the loose mix at 145°C (295°F) for four hours with periodic stirring. Following the short-term oven-aging procedure, compaction of the LMLC and FMLC mixes were performed at the temperatures shown in Table 2.6.

Miv	<b>Compaction Temperature</b>	Temperature Specification Range (°C)					
IVIIX	(° <b>C</b> )						
MB4	$150^{1}$	143–150					
MB15	150	143–150					
MAC	150	143–150					
RAC	145 <sup>2</sup>	143–150					
AR4000	145	_					
Note: Temperatures for MB mixes are from Caltrans Type D-MB specification.							
<sup>1</sup> 293°F	<sup>2</sup> 302°F						

 Table 2.6:
 Compaction Temperatures for LMLC and FMLC

#### 2.4.2 Field-Mixed, Laboratory Compacted Specimens

The field-mixed, laboratory-compacted (FMLC) specimens were prepared using the loose mix collected during construction of the HVS test road. After construction, this material was stored in five-gallon sealed metal cans at room temperature in a warehouse without temperature control for up to several years before compaction. Some further aging may have occurred during the time between site sampling and specimen production. For specimen production, the mix was tested for its maximum specific gravity and compacted following the procedures described above.

The compaction temperatures for field-mixed, lab-compacted specimens were the same as for the LMLC mixes (Table 2.6).

#### 2.4.3 Field-Mixed, Field-Compacted Specimens

The field-mixed, field-compacted (FMFC) specimens were obtained from full-depth cores after the section had been tested, having been subjected to traffic as well as aging. Only the full-depth (90 mm) sections with the MB4 and AR4000 binders provided the 50 mm (2 in) thick cores required for the shear test. The 50 mm test sample was removed from the upper 70 mm (2.7 in) of the full-depth core.

#### 2.5. Ignition Oven Tests

#### 2.5.1 Test Method

California Test CTM382 (Determination of Asphalt Binder Content of Bituminous Mixtures by the Ignition Method) was used to determine binder contents for the field mix collected during construction of the HVS test sections. The ignition oven values were corrected for ignition of the aggregate using aggregate samples also collected during construction. Mixes tested for binder content were RAC-G, MAC15-G, MB15-G, and MB4-G.

#### 2.5.2 Results

Table 2.7 summarizes the results of the ignition oven test on the selected mixes. The results show that the mean field binder contents were approximately 0.5, 0.15, 0.4, and 0.3 percent above the design binder contents for the RAC, MAC15, MB15, and MB4 gap-graded mixes, respectively.

	Design	Ignition Oven	ſ	Test Resi	ılts of Fi	eld Mixe			05%	
Міх Туре	Binder Content <sup>1</sup> (%)	Correction Factor	1	2	3	4	5	Mean	Standard Error	Confidence Interval
RAC-G	8.0	1.86	8.79	8.35	8.54	8.26	-	8.49	0.117	(8.11, 8.86)
MAC15-G	7.4	1.86	7.64	7.42	7.65	7.48	-	7.55	0.058	(7.36, 7.73)
MB15-G	7.1	1.76	7.89	7.66	7.41	7.08	7.58	7.52	0.135	(7.15, 7.90)
MB4-G	7.2	2.15	7.84	7.84	7.62	6.71	-	7.50	0.269	(6.65, 8.36)
<sup>1</sup> Percent by N	<sup>1</sup> Percent by Mass of Aggregate									

 Table 2.7: Summary of Asphalt Contents from Binder Ignition Tests

### **3. BINDER TESTING**

#### **3.1.** Introduction

Binder tests developed by SHRP were performed on the five binders used in this investigation. These tests included the Dynamic Shear Rheometer (DSR) and Bending Beam Rheometer (BBR) tests, and were conducted at the Turner-Fairbanks Highway Research Center of the Federal Highway Administration (FHWA). The tests were conducted on binders in their original condition, after Rolling Thin Film Oven (RTFO) conditioning, and after Pressure Aging Vessel (PAV) conditioning. The tests conducted by the FHWA are identified in AASTHO PG Binder Specification M320.

The AR binder could not be tested by the FHWA because of the size of the rubber particles in the binder. Therefore test results presented in this chapter are for the AR4000, MB4, MB15, and MAC15 binders.

#### 3.2. Dynamic Shear Rheometer

#### 3.2.1 Test Method

AASHTO T315 method was performed to measure the rutting parameter in the AASHTO binder specification (*G/sin* $\delta$ ) and the long-term fatigue performance parameter (*Gsin* $\delta$ ) for the binders.

#### 3.2.2 Results

#### Rutting Criteria

AASHTO M320 defines and places requirements on a rutting parameter for binders,  $G/sin\delta$ , which represents a measure of the contributiuon of the binder to the temperature rutting resistance of the mix. The specification requires that  $G/sin\delta$  must be a minimum of 1.0 kPa for the original asphalt binder and 2.2 kPa after RTFO aging of the binder. Dynamic shear modulus  $G^*$  at 10 rad/s versus test temperatures and the specification requirements for the binders are shown in Figure 3.1 and Figure 3.2 for these two conditions.

#### Fatigue Criterion

In the AASHTO M320 specification, a binder parameter has been introduced to control mix behavior in the intermediate temperature range,  $Gsin\delta$ , and is listed as the fatigue criterion. To mitigate fatigue cracking, the specification requires that  $Gsin\delta$  have a minimum value of 5,000 kPa after PAV aging.
Figure 3.3 illustrates the dynamic shear modulus  $G^*$  at 10 rad/s versus a range of temperatures and contains the specification requirement for  $Gsin\delta$ .



Figure 3.1: G/sinδ summary of DSR test results on original binder.



Figure 3.2: G/sinδ summary of DSR test results on RTFO aged binder.



Figure 3.3: Gsinð summary of DSR test results on PAV-aged binder.

Shear Susceptibility

The Shear Susceptibility of Viscosity (SSV) and Shear Susceptibility of Delta (SSD) are derived from DSR test results and are defined in California Test 381. Reese (5) further developed these parameters for binders used in Type G-modified binder asphalt concrete as follows:

 $SSD \ge 30(0.6+SSV)^3$  for original binder @ 25°C  $SSD \ge -115(SSV) - 50.6$  for PAV-aged binder at @ 25°C

Table 3.1 summarizes the SSD and SSV values for the binders. All the binders satisfy the PAV-aged binder requirement, while only the MB4 binder satisfies the original binder requirement.

#### Test Summary

According to the test results, the ranking of susceptibility of the binders contributing to mix rutting is (from highest susceptibility to lowest):

- 1. MAC15 (binder failed to meet minimum requirements of rutting at any test temperature)
- 2. MB4, MB15
- 3. AR4000

The ranking of susceptibility of the binders contributing to mix fatigue cracking is (from highest to lowest):

- 1. AR4000
- 2. MAC15

- 3. MB15
- 4. MB4

The MB4 and MB15 binders have similar rutting (RTFO aged) and fatigue DSR test results.

Binder	Binder Status*	SSV@25°C	SSD@25°C	Meets Specification for SSD for ORIG	Meets Specification for SSD for PAV						
	ORG	-0.2085	-12.848	No							
AR4000	RTFO	-0.4264	-4.454								
	PAV	-0.2983	-11.428		Yes						
	ORG	-0.4523	6.388	Yes							
MB4	RTFO	-0.4369	2.944								
	PAV	-0.3059	-2.376		Yes						
	ORG	-0.2201	-2.742	No							
MB15	RTFO	-0.2742	-2.013								
	PAV	-0.2490	-5.911		Yes						
	ORG	-0.2289	-0.210	No							
MAC15	RTFO	-0.2585	2.358								
	PAV	-0.2623	-6.898		Yes						
* OF	* ORG: original										
RTFO: rolling thin film oven											
PA	V: pressure	aging vessel.									

Table 3.1: Summary of SSV and SSD Values from DSR Test Results

#### 3.2.3 Master Curves of Shear Complex Modulus

The master curves of the binder shear complex moduli were constructed using time-temperature superposition and a genetic algorithm (3). Figure 3.4 through Figure 3.11 present the  $G^*$  master curves and temperature-shift relationships at various aging conditions for AR4000, MB4, MB15, and MAC15 binders respectively. Observations based on the results of this analysis are:

- For binders aged with the PAV procedure, the complex shear moduli increase across all frequencies for the four binders.
- The MB4 and MB15 binders show small-to-moderate changes among the various aging conditions.
- For original and RTFO aging conditions, the master curves of MAC15 binder are similar; however, the master curve after PAV aging exhibits some deviation from the other two curves. The reason is not clear.
- In general, the temperature-shift relationship does not change significantly for different aging conditions. The only exception is the MAC15 binder. Its temperature sensitivity for the PAV-aged condition increases rapidly at low temperatures and is greater than the temperature sensitivity in the original and RTFO conditions.

Figure 3.12 through Figure 3.14 compare the master curves for the original, RTFO-, and PAV-aging conditions, respectively. In the original and RTFO conditioning, the master curves of MAC15 binder are significantly lower than the master curves for the AR4000, MB4, and MB15 binders.

The ranking of the master curves from greatest to least stiffness for original binder and RTFO conditioning are the same:

- 1. AR4000
- 2. MB4
- 3. MB15
- 4. MAC15

It can be seen in the figures that this ranking is for the middle range of load frequency (also corresponding to middle values of temperature). It can also be seen that at very high and at very low frequencies the AR4000 binder is less stiff than the MB4 and MB15 binders, although it is still stiffer than the MAC15 binder.

The ranking of the master curves for PAV conditioning, from highest stiffness to lowest, is as follows,:

- 1. AR4000
- 2. MAC15
- 3. MB4
- 4. MB15



Figure 3.4: Master curves of shear complex modulus of AR4000.



Figure 3.6: Master curves of shear complex modulus of MB4.



Figure 3.5: Temperature-shift relationships of AR4000.



Figure 3.7: Temperature-shift relationships of MB4.



Figure 3.8: Master curves of shear complex modulus of MB15.



Figure 3.10: Master curves of shear complex modulus of MAC15.



Figure 3.9: Temperature-shift relationships of MB15.



Figure 3.11: Temperature-shift relationships of MAC15.



Figure 3.12: Comparison of G\* Master curves (original).

Figure 3.13: Comparison of G\* Master curves (RTFO).



Figure 3.14: Comparison of G\* Master curves (PAV).

# 3.3. Bending Beam Rheometer

### 3.3.1 Test Method

AASHTO T313 was used to assess the propensity of the binders to develop thermal stresses at low pavement temperatures. The two values obtained from the Bending Beam Rheometer (BBR) are the creep stiffness and the m-value (the rate of change of the creep stiffness versus time of loading). The PG binder specification M320 includes limiting values for these two parameters associated with the low temperature of the PG binder grade. The allowable maximum creep stiffness value is 300 MPa and the minimum m-value is 0.3, both determined at a loading time of 60 seconds.

#### 3.3.2 Results

Table 3.2 lists the temperatures at which creep stiffnesses reached 300 MPa, and m-values reached 0.3. Figure 3.15 and Figure 3.16 show the creep stiffnesses and m-values versus temperature for the un-aged condition and after RTFO and PAV aging. According to the test results and the Superpave specification for thermal cracking, the ranking of susceptibility of the binders to low-temperature thermal cracking from highest susceptibility to lowest is:

- 1. AR4000
- 2. MAC15
- 3. MB15
- 4. MB4

Binder Type	Binder Status*	Temp@S=300 MPa (°C)	Temp@m=0.3 (°C)							
	ORG	-11.5	-15.7							
AR4000	RTFO	-11.3	-15.3							
	PAV	-7.7	-11.3							
	ORG	-20.5	-23.9							
MAC15	RTFO	-18.8	-22.3							
	PAV	-18.0	-19.3							
	ORG	-26.7	-28.3							
MB15	RTFO	-25.6	-26.7							
	PAV	-24.0	-22.5							
	ORG	-31.7	-31.3							
MB4	RTFO	-28.3	-27.8							
	PAV	-25.7	-22.0							
* ORG: original										
RTFO: rolling thin film oven										
PAV: pressure aging vessel										

 Table 3.2: Summary of Bending Beam Rheometer Test Results



Figure 3.15: Creep stiffness summary of BBR test results.



Figure 3.16: Summary of m-value results from BBR tests.

# 4. SHEAR TESTING

## 4.1. Introduction

This chapter provides a summary of the laboratory shear testing study together with the analyses and interpretation of results on field-mixed, laboratory-compacted (FMLC); laboratory-mixed, laboratory-compacted (LMLC); and field-mixed, field-compacted (FMFC) materials. Included are:

- Summary of the stress controlled-deformation Repeated Simple Shear Test (RSST) results
- Identification of the significant factors (or covariates) that affect shear performance
- Discussion of regression models for permanent shear deformation resistance and shear stiffness
- Summary of the collective dataset analysis and regression model

#### 4.1.1 Definitions Used in Statistical Analyses

The factors investigated include:

- Temperature effect (on FMLC material)
- Air-void content effect (on FMLC material)
- Aging effect (on FMLC material)
- Compaction effect (on FMLC and FMFC material)
- Gradation effect (on LMLC material)
- Mix effect (on FMLC and LMLC material)

The response variables are:

- Number of Cycles to 5 Percent Permanent Shear Strain (*Lnpct5*)
- Permanent Shear Strain at 5,000 Cycles (*Lncyc5k*)
- Resilient Shear Modulus, *G*\* (*lng*)

The Resilient Shear Modulus  $(G^*)$  was obtained after 100 repetitions. At this number of repetitions the stress state and temperature, as well as the recoverable shear strain, have stabilized in the shear test specimen. The category covariates and factor levels evaluated include:

- Binder type (*binder*)
  - 1. AR4000 (ar4000)
  - 2. ARB (*rac*)
  - 3. MAC15 (mac15)
  - 4. MB15 (mb15)
  - 5. MB4 (mb4)

- Gradation (grad)
  - 1. Dense-graded (dg)
  - 2. Gap-graded (gg)
- Compaction (*comp*)
  - 1. Field-mixed, laboratory-compacted (*fmlc*)
  - 2. Laboratory-mixed, laboratory-compacted (*lmlc*)
- Conditioning (cond)
  - 1. No conditioning (none)
  - 2. Long-term oven aging for 6 days (aging)
- Air-void content (*av*)
  - 1. 6 percent air-void content (av6)
  - 2. 9 percent air-void content (av9)
- Test temperature (*temp*)
  - 1. 45°C (45C)
  - 2. 55°C (55C)
- Test stress levels (*sts*)
  - 1. 70kPa (sts70)
  - 2. 100kPa (sts100)
  - 3. 130kPa (sts130)

The covariate *binder* has different meanings depending on the test, as follows:

- On all effects other than gradation, *binder* implies a binder type with a specific gradation type and corresponding design binder content as used in this experiment, regardless of mixing method (FMLC or LMLC). It should be noted that AR4000-D and RAC-G are defined by specification and hence a gap-graded mix with the AR4000 binder (AR4000-G) and a dense-graded mix with the rubberized binder (RAC-D) are not included in the experiment. The interpretations of *binder* include:
  - 1. *ar4000:* AR4000 binder with dense-graded gradation and 5.0 percent design asphalt content
  - 2. rac : Asphalt rubber binder with gap gradation and 8.0 percent design asphalt content
  - 3. mac15: MAC15 binder with gap gradation and 7.4 percent design asphalt content
  - 4. *mb15*: MB15 binder with gap gradation and 7.1 percent design asphalt content
  - 5. *mb4:* MB4 binder with gap gradation and 7.2 percent design asphalt content
- When considering the gradation effect, *binder* implies a binder type with a specific design binder content for each gradation. In this instance, *binder* is used as follows:

- 1. *mac15*: MAC15 binder with 6.0 percent asphalt content if dense graded, or 7.4 percent asphalt content if gap graded.
- 2. *mb15:* MB15 binder with 6.0 asphalt content if dense-graded, or 7.1 percent asphalt content if gap graded.
- 3. *mb4:* MB4 binder with 6.3 percent asphalt content if dense graded, or 7.2 percent asphalt content if gap graded.
- When developing the comprehensive regression models with all shear tests, *binder* signifies a binder type with a specific design asphalt content associated with its gradation (dense or gap).

## 4.1.2 Expected Effects of Response Variables on Performance

Expected effects of the response variables from RSST tests are summarized in Table 4.1. The summary of the expected effects is a simplification of a complex distress mechanism, mixture rutting, on unmodified binders. Nonetheless, the summary provides a general guide for interpretation the results presented in this chapter.

The permanent shear strain (PSS) at 5,000 cycles, the test result typically reported from ASHTO T320, is a measure of the material's resistance to permanent shear deformation. Higher values indicate mixes that are weaker, less resistant to permanent shear deformation, and more prone to early rutting failure. Lower values are more desirable as they indicate more resistance to permanent shear deformation and mixture rutting.

The shear resistance is controlled by the entire mix (binder type and stiffness, gradation, aggregate surface texture, compaction, etc.), and not just the binder stiffness. However, for dense-graded mixes with conventional (unmodified) binders, low values are often obtained from stiffer mixes. This relationship between mix stiffness and permanent shear deformation resistance is less clear and often not true for mixes with modified binders.

The number of Cycles to 5 Percent Permanent Shear Strain (PSS) is another measurement of the material's resistance to shear deformation. Here, higher values are desirable as an indication of mixes that are less prone to rut.

The shear stiffness of the material determined after 100 shear load repetitions in the RSST is the third measure of a material's resistance to deformation. Greater shear stiffness often results in lower PSS at 5,000 Cycles and higher Cycles to 5 Percent PSS, although as mentioned previously the relationship

between stiffness and permanent deformation resistance is often contradicted with mixes containing modified binders.

Shear Core Test Response	Distress Mechanism in	Expected Effect on Field Performance					
Variable	Field						
Permanent Shear Strain at	Rutting at high temperatures	Larger shear strain indicates potential early rutting.					
5,000 Cycles							
Cycles to 5 Percent Permanent	Rutting at high temperatures	Low Cycles to 5 Percent Shear Strain indicates					
Shear Strain		potential early rutting.					
Shear Resilient Modulus at	Rutting at high temperatures	High stiffness at high temperatures expected to					
100 <sup>th</sup> Repetition		result in less rutting.					
	Low-temperature cracking	High stiffness at low temperatures expected to					
		increase risk of low-temperature cracking.					

Table 4.1: Summary of Expected Effects of Response Variables from RSST on Performance

## 4.1.3 Presentation of Results

The RSST test results are organized in three sections for each effect:

- Summary boxplots of test results, where each box contains three data points (the three replicates), two of which are the top (highest) and bottom (lowest) sides of the box and one, a white line, that is the middle data point. The height of the box indicates the data variation across the three replicates.
- Identification of significant factors that affect the shear-response variables on an effectcategorized basis.
- Model selection using conventional regression analysis.

In the following discussion, brief explanations of the statistical analyses used in the chapter are provided in Section 4.2, "Temperature Effect." A more detailed discussion of the statistical analyses performed is provided in the detailed first-level fatigue evaluation source report (3). Summary tables of the results of the RSST tests for each mix are attached to this report as Appendix A.

# 4.2. Temperature Effect

This dataset includes the test results of 90 field-mixed, laboratory-compacted (FMLC) cores tested with the following experiment design:

- Five binder types (AR4000, ARB, MAC15, MB15, and MB4)
- One air-void content ( $6.0 \pm 0.5$  percent)
- Two test temperatures (45°C and 55°C)
- Three stress levels (70 kPa, 100 kPa, 130 kPa)
- Three replicates

The covariates investigated were:

- Binder type (*binder*)
- Temperature (*temp*)
- Stress level (*sts*)

#### 4.2.1 Results

Figure 4.1 through Figure 4.3 show boxplots summarizing the RSST test results of temperature effect for Cycles to 5 Percent PSS, PSS at 5,000 Cycles, and Resilient Shear Modulus  $(G^*)$ . The boxplots are categorized by binder/mix type, stress level, and temperature.



Figure 4.1: Summary plots of temperature effect and Cycles to 5 Percent PSS.



Figure 4.2: Summary plots of temperature effect and PSS at 5,000 Cycles.



Figure 4.3: Summary plots of temperature effect and Resilient Shear Modulus, G\*.

The following observations regarding performance implications were made from the plots:

- Temperature has a significant effect on performance. Specimens tested at 55°C consistently show lower Cycles to 5 Percent PSS and higher PSS at 5,000 Cycles than those tested at 45°C for a given stress level.
- There is overlap in shear performance between 45°C and 55°C in that specimens tested at 45°C/130 kPa show lower performance than those tested at 55°C and 70 kPa. There is a definite interdependence of temperature and stress levels.

- For a given temperature, Resilient Shear Modulus  $(G^*)$  stays relatively constant for all stress levels for a given binder type. Shear stiffness, G, at 55°C is without exception lower than  $G^*$  at 45°C for a given binder and is relatively independent of stress level.
- In general, the AR4000-D mix showed the highest resistance to permanent deformation and also had the highest shear stiffnesses. RAC-G mix consistently placed second in resistance to permanent deformation and mix stiffness. MB15-G mix showed the lowest resistance to permanent deformation and the lowest stiffness of the five mixes tested.

The following statistical observations are made from the plots:

- Temperature is highly negatively correlated with Cycles to 5 Percent PSS, particularly at lower stress levels (70 kPa). The effect of temperature diminishes with increased stress level (130 kPa).
- Temperature is highly positively correlated with PSS at 5,000 Cycles.
- Resilient Shear Modulus  $(G^*)$  is highly negatively correlated with temperature.
- All three response variables are significantly affected by the binder/mix types.

## 4.2.2 Identification of Significant Factors

## Correlation Matrix

The correlation matrix (Table 4.2) shows the strength of the linear relationship between the pairs of variables and was used as a quantitative method of identifying significant factors. Correlations that are significant based on an initial threshold correlation of 0.4 are highlighted in the table. The following is observed from the correlation matrix.

- Temperature is negatively correlated with ln *G*\*and ln PSS at 5,000 Cycles. This implies that reduction in test temperature results in increased stiffness and smaller PSS at 5,000 Cycles.
- Stress (*lnsts*) is negatively correlated with PSS at 5,000 Cycles and positively correlated with Cycles to 5 Percent PSS. This implies that increased stress levels results in poorer shear performance and higher rutting potential.
- $G^*$  is positively correlated to PSS at 5,000 Cycles and Cycles to 5 Percent PSS. This implies that increased stiffness results in increased shear performance.

## Analysis of Variance (ANOVA)

The ANOVA results in Table 4.3 provide a second quantitative way to identify significant factors that affect the response variables. The criterion for assessing the importance of effect was set at a 5 percent significance level based on the p-value. Highlighted numbers in the table are considered significant.

	Binder	Air Void	Temperature	Stress	ln G	In Strain at	Ln Cycles
						5K	to 5%
Binder	1.0000	0.0357	0.0072	-0.0522	-0.0338	-0.2511	0.2773
Air Void	0.057	1.0000	-0.04021	-0.0786	0.0462	-0.0174	0.0175
Temp	0.0072	-0.0402	1.0000	-0.0363	-0.6106	-0.3941	0.3525
Stress	-0.0522	-0.0786	-0.03630	1.0000	-0.0279	-0.5569	0.5364
ln G	-0.0338	0.0462	<mark>-0.6106</mark>	-0.0279	1.0000	0.5459	-0.5700
Ln Strain At	-0.2511	-0.0174	<mark>-0.3941</mark>	<mark>-0.5569</mark>	<mark>0.5459</mark>	1.0000	-0.9594
5K							
Ln Cycles	0.2773	0.0175	0.3525	<mark>0.5364</mark>	-0.5700	-0.9594	1.0000
To 5% strain							

 Table 4.2: Correlation Matrix for Temperature Effect

 Table 4.3: Analysis of Variance for Temperature Effect

Coveriete													
Covariate	Df	Sum of Sq	Mean Sq	F Value	p-value								
	Pe	ercent Shear Str	ain at 5000 Cycl	es									
binder	4	178.8680	44.7170	17.3841	<mark>0.0000</mark>								
temp	1	112.8799	112.8799	43.8830	<mark>0.0000</mark>								
sts	1	310.7179	310.7179	120.7941	<mark>0.0000</mark>								
binder:temp	4	46.6564	11.6641	4.5345	<mark>0.0023</mark>								
binder:sts	4	12.1485	3.0371	1.1807	0.3249								
temp:sts	1	0.0136	0.0136	0.0053	0.9421								
Residuals	87	223.7896	2.5723										
Number of Cycles to 5 Percent Shear Strain													
binder	4	8.8771	2.1928	22.0115	<mark>0.0000</mark>								
temp	1	3.2806	3.2806	32.9312	<mark>0.0000</mark>								
sts	1	11.5121	11.5121	115.5589	<mark>0.0000</mark>								
binder:temp	4	1.6123	0.4031	4.0462	<mark>0.0047</mark>								
binder:sts	4	0.4153	0.1038	1.0422	0.3902								
temp:sts	1	0.0362	0.0362	0.3631	0.5483								
Residuals	87	8.667	0.9962										
		Complex Shea	r Modulus G*										
binder	4	23.0896	5.7724	110.3981	<mark>0.0000</mark>								
temp	1	12.2660	12.2660	234.5905	<mark>0.0000</mark>								
sts	1	0.1205	0.1205	2.3040	0.1327								
binder:temp	4	0.4513	0.1128	2.1579	0.0804								
binder:sts	4	0.3924	0.0981	1.8761	0.1218								
temp:sts	1	0.0086	0.0086	0.1650	0.6856								
Residuals	87	4.5489	0.0529										

## **Design** Plots

Design plots are a tool to qualitatively identify significant factors. A series of design plots based on the factor levels used in the study are presented in Figure 4.4 for Cycles to 5 Percent PSS, PSS at 5,000 Cycles, and Resilient Shear Modulus,  $G^*$ . It should be emphasized at the outset that identification of significance using design plots is based on subjective judgment.

The long horizontal bar through the middle of the design plot indicates the overall mean across the entire data set. The smaller horizontal bars identified for each factor level of each variable are the mean value in the data set for that factor level. A larger vertical distance between factor levels means indicates greater significance of that variable.



Figure 4.4: Design plots for temperature effect (6 percent AV).

The following was observed from the design plots:

- Binder type affects all three response variables.
- The difference in shear performance is evident among the various binder/mix types.
- Stress level has the largest effect on PSS at 5,000 Cycles and Cycles to 5 Percent PSS, and no effect on G.

Using the design plots, the PSS at 5,000 Cycles of the various binders is ranked best to worst as follows:

- 1. AR4000-D
- 2. MAC15-G
- 3. RAC-G
- 4. MB4-G
- 5. MB15-G

The ranking of Cycles to 5% PSS for the various binders from best to worst as follows:

- 1. AR4000-D
- 2. RAC-G
- 3. MAC15-G
- 4. MB4-G
- 5. MB15-G

The ranking of Resilient Shear Modulus  $(G^*)$  from stiffest to least stiff:

- 1. AR4000-D
- 2. RAC-G
- 3. MAC15-G
- 4. MB4-G
- 5. MB15-G

Note that the only difference in the rankings for the three performance parameters is in positions 2 and 3, RAC-G and MAC15-G, and that this change only occurs for stiffness and not for the other two parameters.

## Pairs Diagram

A pairs diagram (description presented in Reference 3) was used as a second qualitative method of identifying the significance of correlation. This analysis confirmed the initial observations from the boxplots. (*N.B. An example of a pairs diagram is included subsequently in Figure 4.22.*)

#### Summary of Significant Factors for Temperature Effect

The significant factors were identified from the correlation matrix, analysis of variance, design plot, and other plots (e.g., pairs diagram and interception plots).

Figure 4.5 shows the summary statistics of main effects of the RSST results. If all four criteria show significance in one independent variable, then this is considered as a "very important" factor. If three criteria are check-marked, the factor is considered "important." If only one or two criteria are checked, the variable is considered "less important."

Using this approach for the temperature effect dataset, the following are noted:

• Binder is "important" to all three response variables (Cycles to 5 Percent PSS, PSS at 5,000 Cycles, and G).

- Temperature is "important" to Cycles to 5 Percent PSS and "very important" to both *G*\* and PSS at 5,000 Cycles.
- Applied shear stress is "very important" to PSS at 5,000 Cycles and to Cycles to 5 Percent PSS.

## 4.2.3 Regression Analysis

Mallows'  $C_p$  criterion was used to identify the best subset of covariates for each regression equation, in addition to the analysis of the significance of variables described in much greater detail in Reference 3. The analysis of the significance of explanatory variables included the correlation matrix, analysis of variance, design plots, and pairs diagrams.

#### Cycles to 5 Percent PSS

Binder, temperature, and stress level were the factors most influencing Cycles to 5 Percent PSS. The final model chosen for Cycles to 5 Percent PSS is therefore:

$$E(\ln pct5) = 28.9019 - 0.5919 binder1 - 1.1299 binder2 - 0.3321 binder3 -0.0166 binder4 - 0.2316 temp - 0.0680 sts R2 = 0.72.$$
(4.1)

where Stress (sts) is in kPa and temperature (temp) is in °C.

The term  $E(\ln pct5)$  is the expected value of ln pct5 and the number in parentheses shown under each regression coefficient is the standard error of the estimate of the regression coefficient. The residual standard error is 1.6 on 95 degrees of freedom and the regression line explains as much as 72 percent of the variation in the data.

	Desmonso															(	Cova	riate	s														
Type of Test	Variables		bin	der			gr	ad			cor	np			con	nd			a	c			a	IV			ter	np			st	ss	
	Variables	С	D	F	Α	С	D	F	Α	С	D	F	Α	С	D	F	Α	С	D	F	Α	С	D	F	А	С	D	F	Α	С	D	F	А
Temperature	Lng																									<u>III</u>							
Effect	Lnpct5		$\checkmark$							$\square$		$\square$														MU	MA	M)	<i>MN</i>	<u>IN</u>	M	W)	
(FMLC)	Lnkcyc5		$\checkmark$	$\checkmark$	$\checkmark$					$\square$																	$\checkmark$		$\checkmark$		<i>MA</i>	<i>MA</i>	
Air-Void Content	Lng		$\checkmark$							( )															$\checkmark$	1111	<u>IIII</u>	WA	IIII			[]	
Effect	Lnpct5		$\checkmark$		$\checkmark$					$\square$												WU -	1HII	M	<u>IM</u>	WA WAR	WA WA	ТЙ Т	1111				
(FMLC)	Lnkcyc5		$\checkmark$		$\checkmark$																	NN)	NII)		M	<i>MA</i>	<u>MA</u>	MA W					
	Lng		$\checkmark$														$\checkmark$									1111			1111				
Aging Effect (FMLC)	Lnpct5		$\checkmark$	$\checkmark$						$\square$																<u>IIII</u>	III.	(M)	<u>III</u>				
(I WILC)	Lnkcyc5																										(M)	M)					
Compaction	Lng								$\square$	$\square$	$\checkmark$																	Γ					
Effect	Lnpct5									$\square$	$\checkmark$																						
(FMFC+ FMLC+LMLC)	Lnkcyc5		$\checkmark$	$\checkmark$	$\checkmark$						$\checkmark$	$\checkmark$	$\checkmark$																				
Gradation	Lng	1141		111	))))	1997	1111																										
Effect	Lnpct5	WII.	(M)	1111	W)																												
(LMLC)	Lnkcyc5	MI.		<u>MA</u>	<u>M</u>		$\checkmark$		$\checkmark$																								
Pooled Shear Tests	Lng		$\checkmark$												$\checkmark$		$\checkmark$							$\checkmark$	$\checkmark$		1111						
(FMFC+	Lnpct5		$\checkmark$	$\checkmark$	$\checkmark$		$\checkmark$	$\checkmark$	$\checkmark$	$\square$	$\checkmark$	$\checkmark$			$\checkmark$	$\checkmark$	$\checkmark$							$\checkmark$	$\checkmark$	Ŵ	(MA)	(M)	(M)	Ŵ	M)	M.	M)
FMLC+LMLC)	Lnkcyc5						M				$\checkmark$																$\checkmark$	$\checkmark$	$\checkmark$				
Notes: 1. <i>Lng</i> : G (MPa) in 2. C: correlation m	Notes: 1. <i>Lng</i> : G (MPa) in natural logarithm; <i>Lnpct5</i> : Cycles to 5 Percent PSS (Permanent Shear Strain) in natural logarithm; <i>Lnkcyc5</i> : PSS at 5,000 Cycles in natural logarithm. 2. C: correlation matrix; D: design plot; F: factor plot; A: ANOVA result.																																

3. The four-in-a-row diagonal shaded area means the covariate is "very important" to the corresponding response variable.

Figure 4.5: Summary statistics of main effects.

The term *binder* in the formulation is a category covariate (or factor), which needs to be coded (or parameterized) by "contrasts" for use in the linear regression equation for stiffness. To find the Cycles to 5 Percent PSS for a given mix, the appropriate set of integers shown in Table 4.4 must be used in the equation.

Factor binder: for all the effects equations other than gradation effect equation													
Binder	binder1	binder2	binder3	binder4									
AR4000	-1	-1	-1	-1									
MAC15	1	-1	-1	-1									
MB15	0	2	-1	-1									
MB4	0	0	3	-1									
ARB	0	0	0	4									
Factor binder: for gradation effect equation													
Binder	binder1	binder2											
MAC15	-1	-1											
MB15	1	-1											
MB4	0	2											
Factor cond: for aging effect equation													
Condition	cond												
aging	-1												
none	1												
	Factor comp:	for compaction e	ffect equation										
Compaction	comp1	comp2											
FMFC	-1	-1											
FMLC	1	-1											
LMLC	1	2											
	Factor grad:	for gradation eff	fect equation										
Grading	grad												
DG	-1												
GG	1												

Table 4.4: Contrast Tables of Category Covariates Used in Regression Analyses

#### Example

To determine the regression equation of Cycles to 5 Percent PSS (*ln pct5*) for MB4, the factor values should be set as follows (from Table 4.4):

binder1 = 0, binder2 = 0, binder3 = 3, and binder4 = -1.

This results in the following Cycles to 5 Percent PSS regression equation for the MB4 mixes:

$$E(\ln pct5) = 28.9019 - 0.3321 * 3 - 0.0166 * (-1) - 0.0.2316 temp - 0.0680 sts$$

$$= 27.9222 - 0.2316 temp - 0.0680 sts$$
(4.2)

An analysis of the residuals of the fit was performed, following the procedure described in detail in Reference 3. The results of the analysis showed that there was a slight parabolic trend in the residuals. Inclusion of the interaction term *binder\*temp* would correct this; however, the increasing complexity of the model specification outweighs the increase of  $R^2$ . The assumption of homoscedasticity (same variance in the response variable across the range of explanatory variables) appeared reasonable. The Cook's

distance accompanied with the normal probability plot, the quantile-quantile plot (QQ plot), and the histogram of residuals was used to identify the influential points and possible outliers. The distribution of estimated residuals was found to be close to a normal distribution, which is an assumption of the regression equation. See Figures 4.26 through 4.28 for further details.

#### PSS at 5,000 Cycles

In evaluating the significance of variables affecting the PSS at 5,000 Cycles, stress was identified as "very important," while temperature and binder were identified as "important" in terms of influencing fatigue life. Using Mallows'  $C_p$  criterion, the same factors were identified as the best subset of covariates.

The final model chosen for PSS at 5,000 Cycles is:

$$E(\ln kcyc5) = -6.7512 + 0.1971 binder1 + 0.2637 binder2 + 0.0389 binder3 + 0.0075 binder4 + 0.0413 temp + 0.0128 sts (0.00143) (0.0061) (0.00012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012) (0.0012$$

The residual standard error of the fit is 0.31 on 94 degrees of freedom. The residuals analysis of the PSS at 5,000 Cycles fit showed no significant patterns, indicating that the suggested model is appropriate. And the QQ plot and the histogram both support the methodology as acceptable.

As with the regression equation for Cycles to 5 Percent PSS, the contrast scheme in Table 4.4 should be followed when using the regression equation.

#### <u>Resilient Shear Modulus $(G^*)$ </u>

In evaluating the significance of variables affecting G, temperature was identified as "very important," while binder was identified as "important." Stress level had minimal effect. Using Mallows'  $C_p$  criterion, the same factors were identified as the best subset of covariates.

The final model chosen for  $G^*$  is:

$$E(\ln G) = \underset{(0.2072)}{8.1841} - \underset{(0.0339)}{0.2072} binder1 - \underset{(0.0187)}{0.2330} binder2 - \underset{(0.0139)}{0.00139} binder3 + \underset{(0.0096)}{0.1167} binder4 - \underset{(0.0041)}{0.0041} temp$$

$$R^{2} = 0.90$$
(4.4)

The residual standard error of the fit is 0.21 on 94 degrees of freedom. The residuals analysis of the  $G^*$  fit showed no significant patterns, indicating that the suggested model is appropriate. The QQ plot and the histogram both support the methodology as acceptable.

Analysis procedures similar to those described above are followed in Sections 4.3 through 4.6, and only the results are presented.

# 4.3. Air-Void Content Effect

This test investigated the effect of degree of compaction (the air-void content effect) on shear performance at 45°C and 55°C for various mixes. The experiment design contained a total of 110 tests comprising:

- Five binder types (AR4000, ARB, MAC15, MB15, and MB4),
- Two air-void contents ( $6.0 \pm 0.5$  percent and  $9.0 \pm 1.0$  percent),
  - 1. At  $6.0 \pm 0.5$  percent (90 tests)
    - Three stress levels (70, 100, and 130 kPa), and
    - Three replicates
  - 2. At  $9.0 \pm 1.0$  percent (20 tests)
    - One stress level (70 kPa), and
    - Two replicates

As stated above, the results of 90 tests from the temperature effect study were included in this experiment.

The covariates investigated were primarily:

- Binder type (*binder*)
- Air-void content (*av*)
- Temperature (temp)

Example summary boxplots and design plots are shown in Figure 4.6 through Figure 4.9. The correlation matrices, analyses of variance, and other plots confirm the importance of the listed covariates.



Figure 4.6: Summary boxplots of air-void content effect and Cycles to 5 Percent PSS. (AV6=6 % air-void content; AV9=9% air-void content)



Figure 4.7: Summary boxplots of air-void content and PSS at 5,000 Cycles. (AV6=6 % air-void content; AV9=9% air-void content)



Figure 4.8: Summary boxplots of air-void content and Resilient Shear Modulus (G\*). (AV6=6 % air-void content; AV9=9% air-void content)



Figure 4.9: Design plots of air-void content effect. (AV6=6 % air-void content; AV9=9% air-void content)

A review of the data leads to the following observations:

- Increased air voids yield poorer performance: higher PSS at 5,000 Cycles and lower Cycles to 5 Percent PSS.
- The AR4000 at 55°C and the MAC15-G at 45°C are most effected by changes in air voids, for both PSS at 5,000 Cycles and Cycles to 5 Percent PSS.
- The Resilient Shear Modulus  $(G^*)$  generally decreases with increasing air voids, especially for the RAC-G mix, although not for all mixes or testing conditions.
- For some conditions, resilient modulus remained unchanged.

The final regression models after the identification of significant factors and the iterative procedure of model building are:

$$E(\ln pct5) = 29.8023 - 0.7143 binder1 - 0.9716 binder2 - 0.2487 binder3 - 0.0459 binder4 - 0.8506 av - 0.2387 temp$$
(4.5)

$$R^2 = 0.67$$

$$E(\ln kcyc5) = -6.8575 + 0.2029 \ binder1 + 0.1967 \ binder2 + 0.0688 \ binder3 + 0.0211 \ binder4 + 0.1226 \ av + 0.0456 \ temp$$
(4.6)

$$R^2 = 0.67$$

$$E(\ln G) = 8.5547 - 0.1935 binder1 - 0.2171 binder2 - 0.0774 binder3 + 0.0993 binder4 - 0.0632 av - 0.0700 temp R^{2} = 0.89$$
(4.7)

## 4.4. Aging Effect

This experiment investigated the effect of long-term oven aging (six days at 85°C) on shear performance for the various mixes. The experimental design for the aging tests included:

- Five binder types (AR4000, ARB, MAC15, MB15, and MB4),
- Two temperatures (45°C and 55°C),
- One stress level (70 kPa), and
- Two replicates for a total of four specimens for each mix.

The compacted specimens were conditioned in a forced draft oven for six days at 85°C.

The covariates investigated were:

- Binder type (*binder*)
- Conditioning (*cond*)
- Temperature (*tmp*)

<u>@</u> \_45C 45C -7 55C None ģ ÷ 55C Ln(cycles to 5% PSS) 45C 7 None aging 55C 45C 55C Aging 55C None Aging <u>2</u>. Agi None None Aging \_ ₽ None Aging lone Agin Agin ω **ل** None ging AR4000-D MAC15-G RAC-G **MB15-G** MB4-G

Summary boxplots and design plots are shown in Figure 4.10 through Figure 4.13.

Figure 4.10: Summary boxplots of aging effect and Cycles to 5 Percent PSS. (None=no long-term aging; Aging=6 days long-term oven aging; all specimens at 6% air-voids)



Figure 4.11: Summary boxplots aging effect and PSS at 5,000 Cycles. (None=no long-term aging; Aging=6 days long-term oven aging; all specimens at 6% air-voids)



Figure 4.12: Summary boxplots of aging effect and Resilient Shear Modulus (G\*.) (None=no long-term aging; Aging=6 days long-term oven aging; all specimens at 6% air-voids)



Figure 4.13: Design plots for aging effect. (None=no long-term aging; Aging=6 days long-term oven aging; all specimens at, 6% air-voids)

A review of the data led to the following observations regarding performance and the identification of statistically significant explanatory variables.

- From the summary boxplots, it is apparent that aging results vary widely according to binder/mix type and temperature. The results were inconsistent and appeared unreasonable at times.
  - For AR4000, aging increased the Cycles to 5 Percent PSS and decreased the PSS at 5,000 Cycles, as expected.
  - For the RAC-G and MAC15-G, aging unexpectedly resulted in reduced Cycles to 5 Percent PSS and higher PSS at 5,000 Cycles at both temperatures. At 55°C, aging had little to no effect on Cycles to 5 Percent PSS or PSS at 5,000 Cycles.
  - 3. For MB15-G and MB4-G, the performance showed opposing effects of aging. At 45°C, aging resulted in better performance (more Cycles to 5 Percent PSS and less PSS at 5,000 Cycles) for MB15-G, and the opposite for MB4-G; at 55°C aging resulted in worse performance for MB15-G and better performance for MB4-G.
- From the design plots, it appears that aging is "important" to PSS at 5,000 Cycles and G, and has minimal effect on Cycles to 5 Percent PSS.
- The ranking of the mixes for Cycles to 5 Percent PSS, PSS at 5,000 Cycles, and *G*\* for aged and un-aged tests shows AR4000-D highest and MB15-G lowest, with the other three mixes in various positions in the middle rankings.
- Temperature is very important to the all three measured parameters.

The final regression models after the identification of significant factors and the iterative procedure of model building are:

$$E(\ln pct5) = 22.5692 - 0.9385 binder1 - 0.9770 binder2 - 0.1556 binder3 - 0.1307 binder4 - 0.2010 temp$$
(0.1481) (4.8)

$$R^2 = 0.53$$

$$E(\ln kcyc5) = -5.9345 + 0.2690 binder1 + 0.1877 binder2 + 0.0339 binder3 + 0.0165 binder4 + 0.0422 temp$$
(0.0084) (4.9)

$$R^2 = 0.57$$

$$E(\ln G) = \underset{\substack{(0.2835)\\(0.2835)}}{8.0273} - \underset{\substack{(0.0471)\\(0.0471)}}{0.260} binder 1 - \underset{\substack{(0.2726)\\(0.0251)}}{0.0251} binder 2 - \underset{\substack{(0.1133)\\(0.0190)}}{0.0190} binder 3 + \underset{\substack{(0.0131)\\(0.0303)}}{0.0056} cond - \underset{\substack{(0.0639)\\(0.0056)}}{0.0056} temp$$
(4.10)

 $R^2 = 0.90$ 

# 4.5. Mixing and Compaction Effect

This experiment investigated the effect of mixing and compaction methods on shear performance. The relative experiment design contained a total 118 tests. This experiment used results from tests on field-mixed, laboratory-compacted specimens (FMLC, 90 specimens); laboratory mixed, laboratory-compacted specimens (LMLC, 20 specimens); and field-mixed, field-compacted specimens from the full-depth (90 mm) AR4000 and MB4 sections (FMFC, 6 specimens). Comparisons can only be made for tests with the same testing conditions (45°C and 70 kPa). This resulted in 30 FMLC specimens being included in the experiment. Note that the objective of analysis of the results of this experiment is not mix ranking, but rather to investigate the use of laboratory mixed and compacted materials as opposed to field cores to simulate field performance. The variables in the experiment were:

- Five binder types (AR4000-D, RAC-G, MAC15-G, MB15-G, and MB4-G)
- One temperature  $(45^{\circ}C)$
- One stress level (70 kPa)
- Two and three replicates.

The covariates investigated were primarily:

- Binder type (*binder*)
- Mix and Compaction method (*comp*)

Summary boxplots and design plots are shown in Figure 4.14 through Figure 4.17.



Figure 4.14: Summary boxplots of compaction effect and Cycles to 5 Percent PSS. (6% AV)



Figure 4.15: Summary boxplots of compaction effect and PSS at 5,000 Cycles. (6% AV)



Figure 4.16: Summary boxplots of compaction effect and Resilient Shear Modulus (G\*). (6% AV)



Figure 4.17: Design plots for compaction effect. (6% AV)

A review of the data led to the following observations regarding performance and statistical significance:

- The LMLC specimens had greater PSS at 5,000 Cycles and lower Cycles to 5 Percent PSS than the FMLC.
- The FMFC results were comparable to the LMLC specimens for PSS at 5,000 Cycles and Cycles to 5 Percent PSS.
- The two previous observations indicate that the aging procedures used for the LMLC specimens do a fairly good job of replicating the mixing, aging and compaction of the FMFC specimens, and that the FMLC specimens have more aging than the LMLC and FMFC specimens. This might be expected considering that the FMLC mixes were mixed in the plant, stored in sealed buckets at relatively constant temperature (about 20 °C), and then reheated for laboratory compaction, while the LMLC specimens were mixed from binder that had been stored in sealed containers, short-term oven aged, then immediately compacted.
- The effect of mix and compaction methods was not consistent on the Resilient Shear Modulus  $(G^*)$ . RAC-G is shown to be the stiffest and MB15-G the least stiff.

The final regression models after the identification of significant factors and the iterative procedure of model-building follow. The covariates for both binder type and compaction method are in Table 4.4.

$$E(\ln pct5) = 12.3973 + 1.6453 binder1 - 1.3880 binder2 + 0.0328 binder3 - 0.0366 binder4 + 0.6144 comp1 - 0.4584 comp2$$
(4.11)

$$R^2 = 0.62$$

$$E(\ln kcyc5) = -3.7031 - 0.1471binder1 + 0.2596binder2 + 0.0126binder3 -0.0017binder4 - 0.2266comp1 + 0.0934comp2 (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0$$

 $R^2 = 0.62$ 

$$E(\ln G) = 4.7188 - 0.1396 binder1 - 0.3015 binder2 - 0.0717 binder3 + 0.1443 binder4 + 0.1397 comp1 - 0.1380 comp2 R2 = 0.89$$
(4.13)

## 4.6. Gradation Effect

This experiment investigated the effect of dense- and gap-gradations on Cycles to 5 Percent PSS, PSS at 5,000 Cycles, and G. The experiment design contained a total of twenty-four tests on laboratory-mixed, laboratory compacted cores as follows:

- Three binder types (MAC15, MB15, and MB4),
- Two gradations (dense- and gap-),
- One stress level (70 kPa),
- Two temperatures (45°C and 55°C), and
- One air-void content (6 percent).

There were two replicates for each combination of variables. In addition to these twenty-four cores, four AR4000-D (LMLC) and four RAC-G (LMLC) cores were tested for comparison.

The covariates investigated were primarily:

- Binder type (*binder*)
- Gradation (grad)

Summary boxplots and design plots are shown in Figure 4.18 through Figure 4.21.



Figure 4.18: Summary boxplots of gradation effect and Cycles to 5 Percent PSS. (6% AV)



Figure 4.19: Summary boxplots of gradation effect and PSS at 5,000 Cycles.  $(6\%~{\rm AV})$ 



Figure 4.20: Summary boxplots of gradation effect and Resilient Shear Modulus (G\*).  $(6\%~{\rm AV})$ 



Figure 4.21: Design plots for gradation effect. (6% AV)
A review of the data and box and design plots leads to the following statistical observations:

- From the design plots, gradation has an apparent effect on G, less of an influence on PSS at 5,000 Cycles and little effect on Cycles to 5 Percent PSS.
- For the MAC15 and MB15 binders, the dense gradation resulted in increased Cycles to 5 Percent PSS and higher stiffnesses relative to the gap gradation.
- While dense-graded MB4 mix showed an opposite response with lower Cycles to 5 Percent PSS and higher PSS at 5,000 Cycles, the Resilient Shear Modulus  $(G^*)$  did increase over the gap-gradation.
- MAC15 binder consistently shows the best performance the three modified binders studied, followed by MB4 and MB15.
- In general, these dense-graded mixes are likely to have greater rutting resistance (indicated by PSS results) and reduced risk of rutting of the unbound layers (indicated by greater stiffness), compared to using the same binder in gap-graded mixes.
- The reason for the poor results for the AR4000 relative to the other mixes in the gradation study is not certain, considering that it had the best results for the field–mixed, laboratory-compacted specimens. The difference may be due to differences in aging in the laboratory compared to the plant during mixing.

The final regression models after the identification of significant factors and the iterative procedure of model building follow. The covariates for both binder type and compaction method are in Table 4.4.

$$E(\ln pct5) = 11.7599 - 2.2128 binder1 - 0.4199 binder2 - 0.2848 grad$$

$$R^{2} = 0.83$$
(4.14)

$$E(\ln kcyc5) = -3.7107 + 0.3713 binder1 + 0.0812 binder2 + 0.1556 grad$$

$$R^{2} = 0.86$$
(4.15)

$$E(\ln G) = 4.4254 - \underbrace{0.3533binder1 - 0.0007binder2 - 0.2422grad}_{(0.0516)} grad$$

$$R2 = 0.77$$
(4.16)

## 4.7. Pooled Shear Tests

Analyses of grouped results were undertaken to develop comprehensive models that better describe the rutting performance of the materials tested. The dataset used consisted of all shear tests. The covariates inspected were:

- Binder type (binder)
- Gradation (grad)
- Compaction (*comp*)
- Aging (cond)
- Air-void content (*av*)
- Temperature (*temp*)
- Stress (*sts*)

Figure 4.22 presents the pairs diagram showing the relationships between variables. Summary boxplots and design plots are shown in Figure 4.23 through Figure 4.25.



Figure 4.22: Pairs diagram showing relationships among variables.



Figure 4.23: Summary boxplots of pooled shear testing for permanent shear strain at cycles. (6% AV)



Figure 4.24: Summary boxplots of pooled shear testing for Cycles to 5 Percent PSS.  $(6\%~{\rm AV})$ 



Figure 4.25: Example design plots for pooled shear tests.

A review of the data and charts led to the following observations:

- All five binders show consistent performance for cycles to 2 and 5 percent PSS and PSS at 1,000, 2,000, and 5,000 cycles.
- AR4000-D consistently shows the best rutting performance in these tests.
- MB15 consistently showed the poorest rutting performance.
- Stress level, temperature, and gradation are "very important" to Cycles to 5 Percent PSS, PSS at 5,000 Cycles, and G.
- Air-void content, aging, and compaction method are less important to these same parameters.
- Both the degree of compaction and the applied stress level have little influence on Resilient Shear Modulus (*G*<sup>\*</sup>).

The final regression models for grouped shear testing after the identification of significant factors and the iterative procedure of model building are:

$$E(\ln pct5) = 32.6002 + 0.2724 \ binder1 - 0.6457 \ binder2 - 0.0650 \ binder3 \\ + 0.1139 \ binder4 + 0.7332 \ comp1 - 0.5516 \ comp2 - 0.9076 \ grad \\ - 0.7721 \ av - 0.2321 \ temp - 0.0658 \ sts \\ R^2 = 0.64$$

$$(4.17)$$

$$E(\ln kcyc5) = -7.4491 - 0.4308 binder1 - 0.0218 binder2 - 0.0574 binder3 -0.0562 binder4 - 0.2470 comp1 + 0.1725 comp2 + 0.4810 grad(0.0169) + 0.1365 av + 0.0384 temp + 0.0130 sts + 0.0477 comp1 \cdot grad(0.0255) - 0.0384 temp + 0.0130 sts + 0.0477 comp1 \cdot grad-0.1513 comp2 \cdot grad
$$R^{2} = 0.65$$$$

$$E(\ln G) = \frac{8.2761 - 0.0047}{(0.0272)} \frac{binder1 - 0.1508 binder2 - 0.0391 binder3}{(0.0147)} + \frac{0.1483 binder4 + 0.1365 comp1 - 0.1150}{(0.0544)} \frac{comp2 - 0.1706 cond}{(0.0311)} - \frac{0.3011}{(0.0187)} \frac{grad}{ar} - \frac{0.0490}{(0.0187)} \frac{av}{av} - \frac{0.0654}{(0.0041)} temp$$

$$R^{2} = 0.84$$
(4.19)

Figure 4.26 through Figure 4.28 present the residual plots for the regression analysis of pooled shear tests.



Figure 4.26: Residual plots of Ln(pct5). (Pooled Shear Tests)



Figure 4.27: Residual plots of Ln(kcyc5) (Pooled Shear Tests).



Figure 4.28: Residual plots of LnG. (Pooled Shear Tests).

## 4.8. Summary of Factor Identification

A main-effect summary table (Figure 4.5) was developed based on quantitative and qualitative analyses to identify the significant factors of the study. This table describes the four analyses performed in this study.

In the table, the more blocks checked and cross-hatched for a variable, the more important or significant it is. Although this process is somewhat subjective, the factors identified as significant closely match the covariates selected for regression analysis using Mallow's  $C_p$  criterion. A summary of the factor identification process includes:

- The temperature effect on shear performance is apparent and very important to all three response variables: Cycles to 5 Percent PSS, PSS at 5,000 Cycles, and G.
- The air-void content effect is most significant for Cycles to 5 Percent PSS and PSS at 5,000 Cycles.
- The aging effect is important for  $G^*$  and only somewhat important for Cycles to 5 Percent PSS.
- The compaction method is somewhat important for all the response variables.
- The gradation effect is significant for mix stiffness G. Dense-graded mixes showed higher stiffnesses than the gap-graded mixes.
- Binder type has a distinct effect on all the response variables.

## 4.9. Summary of Regression Analysis

The regression models are summarized in Tables 4.5 through 4.7. Given that a partial factorial experimental design was followed, extrapolations or inferences of model predictions beyond the data range should be undertaken with caution.

Test	Regression Model	$\mathbf{R}^2$
Temperature Effect (FMLC)	$E(\ln pct5) = 28.9019 - 0.5919 binder1 - 1.1299 binder2 - 0.3321 binder3 - 0.0166 binder4 - 0.2316 temp - 0.0680 sts$	0.72
Air-void Content Effect (FMLC)	$E(\ln pct5) = 29.8023 - 0.7143 binder1 - 0.9716 binder2 - 0.2487 binder3 - 0.0459 binder4 - 0.8506 av - 0.2387 temp _{(0.0412)}(0.0412)$	0.67
Aging Effect (FMLC)	$E(\ln pct5) = 22.5692 - 0.9385 binder1 - 0.9770 binder2 - 0.1556 binder3 - 0.1307 binder4 - 0.2010 temp_{(0.0439)}$	0.53
Compaction Effect (FMFC+FMLC+ LMLC)	$E(\ln pct5) = 12.3973 + 1.6453 binder1 - 1.3880 binder2 + 0.0328 binder3 - 0.0366 binder4 + 0.6144 comp1 - 0.4584 comp2 _{(0.1311)} - 0.4584 comp2 \\_{(0.04773)} - 0.04584 comp2 \\_{(0.0$	0.62
Gradation Effect (LMLC)	$E(\ln pct5) = 11.7599 - 2.2128 binder1 - 0.4199 binder2 - 0.2848 grad$	0.86
Pooled Shear Tests (FMFC+FMLC+ LMLC)	$E(\ln pct5) = 32.6002 + 0.2724 binder1 - 0.6457 binder2 - 0.0650 binder3 + 0.1139 binder4 + 0.7332 comp1 - 0.5516 comp2 - 0.9076 grad (0.03965) - 0.7721 av - 0.2321 temp - 0.0658 sts (0.1280) - 0.7721 av - 0.2321 temp - 0.0658 sts (0.0060) - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - 0.0000 - $	0.64

Fable 4.5:    Re	gression Models fo	r In Cycles to	5 Percent P	ermanent Shear	Strain
1 4010 1101 110	SI COSTOIL TITOUCIO IO	i m Cycles to		ci munche Sheur	Julain

Test	Regression Model	$\mathbf{R}^2$
Temperature Effect (FMLC)	$E(\ln kcyc5) = -6.7512 + 0.1971binder1 + 0.2637binder2 + 0.0389binder3 + 0.0075binder4 + 0.0413temp + 0.0128sts$	0.74
Air-void Content Effect (FMLC)	$E(\ln kcyc5) = -6.8575 + 0.2029 binder1 + 0.1967 binder2 + 0.0688 binder3 + 0.0211 binder4 + 0.1226 av + 0.0456 temp (0.0179) (0.0278) (0.0079) (0.0079) (0.0079)$	0.67
Aging Effect (FMLC)	$E(\ln kcyc5) = -5.9345 + 0.2690 binder1 + 0.1877 binder2 + 0.0339 binder3 + 0.0165 binder4 + 0.0422 temp_{(0.0084)}$	0.57
Compaction Effect (FMFC+FMLC+ LMLC)	$E(\ln kcyc5) = -3.7031 - 0.1471 binder1 + 0.2596 binder2 + 0.0126 binder3 -0.0017 binder4 - 0.2266 comp1 + 0.0934 comp2 (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377) (0.0377$	0.62
Gradation Effect (LMLC)	$E(\ln kcyc5) = -3.7107 + 0.3713 binder1 + 0.0812 binder2 + 0.1556 grad$	0.83
Pooled Shear Tests (FMFC+FMLC+ LMLC)	$E(\ln kcyc5) = -7.4491 - 0.4308 binder1 - 0.0218 binder2 - 0.0574 binder3 -0.0562 binder4 - 0.2470 comp1 + 0.1725 comp2 + 0.4810 grad (0.0169) + 0.1365 av + 0.0384 temp + 0.0130 sts + 0.0477 comp1 \cdot grad -0.1513 comp2 \cdot grad -0.1513 comp2 \cdot grad$	0.65

 Table 4.6: Regression Models for In Permanent Shear Strain at 5,000 Cycles

 Table 4.7: Regression Models for Resilient Shear Modulus (G\*)

Test	Regression Model	$\mathbf{R}^2$
Temperature Effect (FMLC)	$E(\ln G) = \underset{(0.2072)}{8.1841} - \underset{(0.0339)}{0.0339} binder1 - \underset{(0.0187)}{0.2330} binder2 - \underset{(0.0139)}{0.00187} binder3 + \underset{(0.0096)}{0.0139} binder4 - \underset{(0.0041)}{0.0041} binder3$	0.90
Air-void Content Effect (FMLC)	$E(\ln G) = 8.5547 - 0.1935 binder1 - 0.2171 binder2 - 0.0774 binder3 + 0.0993 binder4 - 0.0632 av - 0.0700 temp _{(0.0123)} (0.0187) (0.0187) (0.0054)$	0.89
Aging Effect (FMLC)	$E(\ln G) = 8.0273 - 0.2669 binder1 - 0.2726 binder2 - 0.1133 binder3 + 0.1260 binder4 - 0.1648 cond - 0.0639 temp _{(0.0131)} (0.0303) (0.0303) (0.0056)$	0.90
Compaction Effect (FMFC+FMLC+ LMLC)	$E(\ln G) = 4.7188 - 0.1396 binder1 - 0.3015 binder2 - 0.0717 binder3 + 0.1443 binder4 + 0.1397 comp1 - 0.1380 comp2 _{(0.0154)} (0.0518) (0.0236)$	0.89
Gradation Effect (LMLC)	$E(\ln G) = 4.4254 - \underbrace{0.3533binder1 - 0.0007binder2 - 0.2422grad}_{(0.0505)} grad$	0.77
Pooled Shear Tests (FMFC+FMLC+ LMLC)	$E(\ln G) = 8.2761 - \underbrace{0.0047}_{(0.0592)} binder1 - \underbrace{0.1508}_{(0.0237)} binder2 - \underbrace{0.0391}_{(0.0147)} binder3 + \underbrace{0.1483}_{(0.0113)} binder4 + \underbrace{0.1365}_{(0.0544)} comp1 - \underbrace{0.1150}_{(0.0283)} comp2 - \underbrace{0.1706}_{(0.0311)} cond + \underbrace{0.3011}_{(0.0531)} grad - \underbrace{0.0490}_{(0.0187)} av - \underbrace{0.0654}_{(0.0041)} temp$	0.84

## 4.9.1 Cycles to 5 Percent Permanent Shear Strain

Figure 4.29 and Figure 4.30 schematically summarize the regression models listed in Table 4.5. Evaluations of these data indicate the following:

- The ranking of Cycles to 5 Percent PSS of the binders from best performance to poorest under all effects (except compaction) is in the order listed below for the FMLC specimens.
  - 1. AR4000-D
  - 2. MAC15-G
  - 3. RAC-G
  - 4. MB4-G
  - 5. MB15-G
- This order changes only for compaction effect as follows:
  - 1. MAC15-G
  - 2. MB4-G
  - 3. AR4000-D
  - 4. RAC-G
  - 5. MB15-G

## 4.9.2 **PSS at 5,000 Cycles**

Figure 4.31 and Figure 4.32 schematically summarize the regression models for PSS at 5,000 Cycles using the regression equations shown in Table 4.6. The following are observed, which is consistent with Cycles to 5 Percent PSS:

- The ranking of PSS at 5,000 Cycles for the binders from best performance to worst under all effects (except compaction) is in the order listed below for the FMLC specimens.
  - 1. AR4000-D
  - 2. MAC15-G
  - 3. RAC-G
  - 4. MB4-G
  - 5. MB15-G
- This order changes only for compaction effect as follows:
  - 1. MAC15-G
  - 2. AR4000-D
  - 3. RAC-G
  - 4. MB4-G
  - 5. MB15-G

## **4.9.3** Resilient Shear Modulus $(G^*)$

Figure 4.33 and Figure 4.34 schematically summarize the regression models for Resilient Shear Modulus  $(G^*)$  as in Table 4.7. The following are observed:

- The ranking of *G*<sup>\*</sup> for the binders from stiffest to least stiff under all effects (except compaction) is in the order listed below for the FMLC specimens.
  - 1. AR4000-D
  - 2. RAC-G
  - 3. MAC15-G
  - 4. MB4-G
  - 5. MB15-G
- This order changes only for compaction effect as follows:
  - 1. RAC-G
  - 2. AR4000-D
  - 3. MAC15-G
  - 4. MB4-G
  - 5. MB15-G



Figure 4.29: Schematic summary of Cycles to 5 Percent PSS regression models – Part A.



Figure 4.30: Schematic summary of Cycles to 5 Percent PSS regression models – Part B.



Figure 4.31: Schematic summary of PSS at 5,000 Cycles regression models – Part A.



Figure 4.32: Schematic summary of PSS at 5,000 Cycles regression models – Part B.



Figure 4.33: Schematic summary of Resilient Shear Modulus (G\*) regression models – Part A.



Figure 4.34: Schematic summary of Resilient Shear Modulus (G\*) regression models – Part B.

In Figure 4.29 through Figure 4.34 (a) through (f) results were calculated using the regression equation from the associated experiment, as shown in Table 4.5 through Table 4.7, and (g) and (h) results were calculated using the pooled shear test results equations in the tables.

#### 4.10. Weibull Analysis

Two-stage Weibull analysis was briefly studied on selected shear test results. The data can be plotted in terms of the double natural logarithm ln (ln taken twice) of the PSS versus the number of shear repetitions in the test. A more detailed analysis of the Weibull parameters for all of the core test results will be included in the second-level analysis report to follow. Potentially, a three-stage Weibull model may be discussed.

When plotted as a double log of PSS at 5,000 Cycles versus log of Cycles to 5 Percent PSS, the deformation curve for a shear permanent deformation test (RSST) typically consists of two (Figure 4.35 and Figure 4.36) or three stages (to be discussed in the second-level report), namely:

- 1. Stage I, an initial stage;
- 2. Stage II, deformation, during which there is a steady rate of shear deformation versus repetitions; and

3. Stage III, potential tertiary flow, during which the rate of shear deformation versus repetitions is greater than in Stage II.



Figure 4.35: Two-Stage Weibull fitting for RSST-CH test result.

These three stages can be analyzed using a three-stage Weibull equation (3, 5); in this study it was used to evaluate each shear test. The associated shear parameters that define the three-stage Weibull fatigue curve are:

- Six parameters taken directly from the curve:  $\ln \alpha_1$ ,  $\beta_1$ ,  $\ln \alpha_2$ ,  $\beta_2$ ,  $\ln \alpha_3$ ,  $\beta_3$ ; and
- Two parameters derived from the curve, i.e., the repetitions at which the transitions between Stages I and II, and Stage II and Stage III occur: *n*<sub>1</sub>, *n*<sub>2</sub>, respectively.

As shown in Figure 4.37, the Ln(PSS at 5,000 Cycles) and the Ln(Cycles to 5 Percent PSS) are highly negatively-correlated with correlation = -0.951. It might imply that for *ranking mix performance*, for mixes with polymer-modified binders or for mixes that take more than 5,000 cycles to reach 5 percent permanent shear strain, the RSST can be stopped at 5,000 cycles as specified in AASHTO T320.



Figure 4.36: PSS Correlation.

$$\begin{cases} PSS = \exp(-\alpha_1 n^{\beta_1}) & , 0 < n \le n_0 \\ PSS = \exp(-\alpha_2 (n-\gamma)^{\beta_2}) & , n > n_0 \end{cases}$$
(4.20)

where PSS is the Permanent Shear Strain and n is number of loading repetitions.

Fitting results:

 $\alpha_1 = \exp(2.075632) = 7.9696$   $\alpha_2 = \exp(2.039126) = 7.6839$   $\beta_1 = -0.0832$   $\beta_2 = -0.0779$   $\gamma = 23.73$  $n_0 = 372.59$ 



Figure 4.37: Correlation between PSS at 5,000 Cycles and Cycles to 5 Percent PSS.

## 4.11. Second-Level Analysis

Second-level analysis reports will be prepared on completion of HVS testing, laboratory testing, and forensic evaluations. These reports will include:

- As-built layer thicknesses of the HVS sections;
- Backcalculation of moduli from deflection measurements (RSD, MDD, and FWD);
- Verification of data collected from in-depth measurements with visual observations from test pits;
- Comparison of performance between HVS test sections;
- Comparison of performance between HVS test sections, after accounting for any differences in underlying support conditions;
- Comparisons of HVS test results with laboratory test results;
- Analysis of expected shear performance for a range of pavement structures containing densegraded mixes with MB4, MB15, and MAC15 binders, and comparison with dense-graded mixes containing conventional and other modified binders;
- Analysis of the shear test results using a two- or three-stage Weibull analysis; and
- Final recommendations.

## 5. CONCLUSIONS

This summary report is one of a series of reports detailing the results of laboratory testing undertaken in conjunction with HVS testing to validate Caltrans overlay strategies for the overlay of cracked asphalt pavements. The report describes the results of the laboratory shear test study, carried out on five binders (AR4000, asphalt rubber, and three modified binders, MB4, MB4 with 15 percent recycled rubber (referred to as MB15), and MAC15TR (referred to as MAC15). The AR4000 was tested in dense-graded mixes only, and the asphalt rubber was tested only in gap-graded mixes (RAC-G). The MB4, MB15, and MAC15 were tested in both gap-graded and dense-graded mixes.

The Repeated Simple Shear Test (RSST) at Constant Height, AASHTO M320 Procedure C, was the only test used in this study. Across several experiments tests were performed at two temperatures, 45°C and 55°C (113°F and 131°F); three stress levels, two air-void contents, and with and without six days of long-term oven aging.

Test specimens were prepared from field mix compacted in the laboratory, mix prepared and compacted in the laboratory from aggregate and binder samples taken during construction of the HVS test sections, and field cores.

Binder tests were performed by the Federal Highway Administraton (FHWA).

Comparison of the laboratory and test section performance, including the results of a forensic investigation to be conducted when testing is complete, will be discussed in a second-level reports once the data from all of the studies has been collected. Findings and observations based on the laboratory shear study are discussed below. It should be noted that the study was focused on the use of modified binders in thin overlays on existing cracked asphalt surfaces, and not in structural layers.

#### 5.1. Findings and Observations

#### Summary of Binder Tests

• Based on Bending Beam Rheometer (BBR) results from tests performed by the FHWA, the ranking of propensity to low temperature thermal cracking is listed below, from best to worst. Asphalt rubber binder was not tested.

- 1. MB4
- 2. MB15
- 3. MAC15
- 4. AR4000
- The order of thermal cracking potential is closely matched with the order of initial stiffness in the fatigue beam tests and flexural frequency sweep results; hence a mix with a higher initial stiffness might have a higher thermal cracking potential.
- The Dynamic Shear Rheometer (DSR) test results obtained from the FHWA indicated that:
  - MAC15 failed to meet the Superpave rutting specification.
  - MB4 and MB15 binders have better rutting resistance capacities than the AR4000 binder.
  - According to the Superpave specification, the ranking of fatigue resistance capacity is in the order listed below, from best to worst, which is the same ranking obtained for initial stiffness during laboratory mix fatigue tests, from highest to lowest stiffness.
    - 1. MB4
    - 2. MB15
    - 3. MAC15
    - 4. AR4000

#### **Overall Summary of Repeated Simple Shear Test Results**

- The binder type has an overall effect on all the response variables including permanent shear strain (PSS) at 5,000 Cycles, Cycles to 5 Percent PSS, and shear stiffness (G). As expected, the temperature effect on all three response variables is immediately apparent and significant. The other effects assessed (for comparison with HVS testing) reveal that:
  - Air-void content had a significant effect for Cycles to 5 Percent PSS and PSS at 5,000 Cycles but the effect was not significant for G.
  - Overall, the long-term aging effect is only minimally significant.
  - For MAC15, MB15, and MB4 mixes, all the response variables are significantly affected by the change from a gap gradation to a dense gradation, with expected rutting performance improved when the binders are used with the dense gradation as opposed to gap gradation.

#### Ranking of PSS at 5,000 Cycles, Cycles to 5 Percent PSS, and Shear Resilient modulus (G)

The ranking of these parameters under the various specimen preparation and testing conditions for RSST test used in this study is listed below from best to worst with respect to expected rutting performance.

Cycles to 5 Percent PSS	PSS 5,000 Cycles	Resilient Shear Modulus $(G^*)$
1. AR4000-D	1. AR4000-D	1. AR4000-D
2. MAC15-G	2. MAC15-G	2. RAC-G
3. RAC-G	3. RAC-G	3. MAC15-G
4. MB4-G	4. MB4-G	4. MB4-G
5. MB15-G	5. MB15-G	5. MB15-G

### Dense-Graded versus Gap-Graded Mixes

- The optimum binder contents used in the mix designs based on Hveem stabilometer tests for the MAC15, MB15, and MB4 dense-graded mixes (6.0, 6.0, and 6.3 percent respectively) were lower than the optimum binder contents used in the mix designs of the gap-graded mixes (7.4, 7.1, and 7.2 percent respectively), with all mix designs performed following standard Caltrans methods.
- Limited shear testing of modified binders in dense-graded mixes led to the following observation, based on testing of laboratory-mixed, laboratory-compacted specimens:
  - The PSS at 5,000 Cycles and Cycles to 5 Percent PSS results for the dense-graded mixes indicated generally better rutting performance than those of the corresponding gap-graded mixes.

## 5.2. Recommendations

No recommendations as to the use of modified binder mixes are made at this time. These recommendations will be included in the second-level analysis report that will be prepared and submitted on completion of all HVS and laboratory testing.

## 6. **REFERENCES**

- Generic Experimental Design for Product/Strategy Evaluation—Crumb Rubber Modified Materials. 2005. State of California Department of Transportation (Caltrans), Sacramento, CA: Materials and Engineering Testing Services, Office of Flexible Pavement Materials. 55 pp.
- Reflective Cracking Study: Workplan for the Comparison of MB, RAC-G, and DGAC Mixes Under HVS and Laboratory Testing. 2003. Davis and Berkeley, CA: University of California Pavement Research Center. (UCPRC-WP-2003-01).
- TSAI, B.W., Jones, D., Harvey, J. and Monismith, C. 2006. Reflective Cracking Study: Firstlevel Report on Laboratory Fatigue Testing. Davis and Berkeley, CA: University of California Pavement Research Center. (UCPRC-RR-2006-08).
- BELL, C.A. and Sosnovske, D. 1994. Aging: Binder Validation. Washington, D.C.: Strategic Highway Research Program, National Research Council. (SHRP-A-384).
- TSAI, B.W., Harvey, J. and Monismith, C. 2005. Characterization of Mix Fatigue Damage Process Using a Three-Stage Weibull Equation and a Tree-Based Model. In Transportation Research Record 1929. Washington, D.C.: Transportation Research Board. pp. 227-237.
- REESE, R.E. 1997. Properties of Aged Binder Related to Asphalt Concrete Fatigue Life. *Journal* Association of Asphalt Paving Technologists, Vol. 66, pp. 604-632.

## **APPENDIX A: SUMMARY OF RESULTS**

Table A.1: Summary of shear laboratory test results for AR4000 mixes (Temperature effect).
Table A.2: Summary of shear laboratory test results for RAC mixes (Temperature effect).
Table A.3: Summary of shear laboratory test results for MAC15 mixes (Temperature effect).
Table A.4: Summary of shear laboratory test results for MB15 mixes (Temperature effect).
Table A.5: Summary of shear laboratory test results for MB4 mixes (Temperature effect).
Table A.6: Summary of shear laboratory test results for all mixes (Air-Void Content Effect)
Table A.7: Summary of shear laboratory test results for all mixes (Aging Effect).
Table A.8: Summary of shear laboratory test results for all mixes (Compaction Effect).
Table A.8: Summary of shear laboratory test results for all mixes (Compaction Effect).

Specimen ID	Binder Type	Comp.	Grad.	AV (%)	AC (%)	Test Stress Level (kPa)	Test Temp. (°C)	G (MPa)	n@5% PSS	PSS@ 5000cycles
DGAC-6-6-1-7045	AR-4000	FMLC	DG	5.8	5.0	70	45	173	46,078,399	0.010270
DGAC-6-14-3-7045	AR-4000	FMLC	DG	5.8	5.0	70	45	249	6,222,167	0.010514
DGAC-6-15-2-7045	AR-4000	FMLC	DG	6.4	5.0	70	45	268	1,058,305	0.016905
DGAC-6-10-2-7055	AR-4000	FMLC	DG	5.8	5.0	70	55	113	3,570,699	0.012686
DGAC-6-10-3-7055	AR-4000	FMLC	DG	6.5	5.0	70	55	111	251,616	0.017752
DGAC-6-11-1-7055	AR-4000	FMLC	DG	5.9	5.0	70	55	95	272,762	0.020137
DGAC-6-14-1-10045	AR-4000	FMLC	DG	6.5	5.0	100	45	222	10,110,764	0.010332
DGAC-6-14-2-10045	AR-4000	FMLC	DG	6.0	5.0	100	45	231	9,107,224	0.011206
DGAC-6-15-3-10045	AR-4000	FMLC	DG	5.9	5.0	100	45	393	25,944	0.026597
DGAC-6-16-1-10045	AR-4000	FMLC	DG	6.0	5.0	100	45	282	2,915,617	0.013628
DGAC-6-16-3-10045	AR-4000	FMLC	DG	6.1	5.0	100	45	342	5,746,161	0.010848
DGAC-6-8-1-10055	AR-4000	FMLC	DG	6.0	5.0	100	55	123	49,276	0.025691
DGAC-6-8-2-10055	AR-4000	FMLC	DG	6.0	5.0	100	55	126	86,808	0.022751
DGAC-6-8-3-10055	AR-4000	FMLC	DG	6.5	5.0	100	55	115	9,047	0.037099
DGAC-6-10-1-13045	AR-4000	FMLC	DG	5.9	5.0	130	45	296	418,300	0.017335
DGAC-6-11-2-13045	AR-4000	FMLC	DG	5.9	5.0	130	45	208	25,790	0.033128
DGAC-6-11-3-13045	AR-4000	FMLC	DG	5.8	5.0	130	45	268	164,903	0.021585
DGAC-6-4-1-13055	AR-4000	FMLC	DG	6.5	5.0	130	55	114	5,083	0.049719
DGAC-6-6-2-13055	AR-4000	FMLC	DG	6.1	5.0	130	55	101	8,105	0.040761
DGAC-6-6-3-13055	AR-4000	FMLC	DG	6.3	5.0	130	55	123	27,866	0.029691
Note:	•			•	•	•	•	•	•	•

Table A.1: Summary of Shear Laboratory Test Results for AR4000 Mixes (Temperature Effect).

 $(FMLC, AV = 6. \pm 0.5\%, AC = 5.0\%)$ 

FMLC: field-mixed laboratory-compacted; LMLC: laboratory-mixed laboratory compacted.
 DG: dense-graded; GG:gap-graded.

Specimen ID	Binder Type	Comp.	Grad.	AV (%)	AC (%)	Test Stress Level (kPa)	Test Temp. (°C)	G (MPa)	n@5% PSS	PSS@ 5000 cycles
RACG-6-10-3-7045	ARB	FMLC	GG	6.9	8.0	70	45	278	59,533	0.026025
RACG-6-13-1-7045	ARB	FMLC	GG	6.0	8.0	70	45	340	924,374	0.013866
RACG-6-16-1-7045	ARB	FMLC	GG	6.0	8.0	70	45	262	19,000,787	0.012036
RACG-6-21-2-7045	ARB	FMLC	GG	6.5	8.0	70	45	224	408,923	0.023341
RACG-6-22-3-7045	ARB	FMLC	GG	5.9	8.0	70	45	231	51,876	0.029105
RACG-6-9-3-7045	ARB	FMLC	GG	6.4	8.0	70	45	218	1,461,806	0.013873
RACG-6-13-2-7055	ARB	FMLC	GG	6.5	8.0	70	55	156	116,446	0.027573
RACG-6-16-2-7055	ARB	FMLC	GG	5.7	8.0	70	55	95	21,369	0.042460
RACG-6-18-1-7055	ARB	FMLC	GG	5.8	8.0	70	55	100	161,947	0.027892
RACG-6-20-2-7055	ARB	FMLC	GG	5.3	8.0	70	55	151	82,473	0.026392
RACG-6-22-1-7055	ARB	FMLC	GG	6.4	8.0	70	55	92	70,552	0.036577
RACG-6-19-1-10045	ARB	FMLC	GG	6.4	8.0	100	45	256	399,546	0.023893
RACG-6-6-1-10045	ARB	FMLC	GG	6.5	8.0	100	45	177	6,043	0.048453
RACG-6-6-2-10045	ARB	FMLC	GG	6.4	8.0	100	45	251	6,000	0.047912
RACG-6-17-3-10055	ARB	FMLC	GG	5.7	8.0	100	55	102	70,057	0.031987
RACG-6-18-2-10055	ARB	FMLC	GG	6.3	8.0	100	55	82	4,392	0.051036
RACG-6-22-2-10055	ARB	FMLC	GG	5.9	8.0	100	55	122	4,804	0.050382
RACG-6-17-1-13045	ARB	FMLC	GG	6.4	8.0	130	45	183	115,428	0.028685
RACG-6-20-1-13045	ARB	FMLC	GG	5.8	8.0	130	45	287	43,624	0.029713
RACG-6-20-2-13045	ARB	FMLC	GG	5.5	8.0	130	45	272	4,546	0.050950
RACG-6-21-3-13045	ARB	FMLC	GG	6.4	8.0	130	45	206	25,913	0.038179
RACG-6-17-2-13055	ARB	FMLC	GG	6.3	8.0	130	55	153	16,605	0.036441
RACG-6-18-3-13055	ARB	FMLC	GG	6.1	8.0	130	55	83	1,482	0.061992
RACG-6-19-3-13055	ARB	FMLC	GG	6.2	8.0	130	55	105	13,842	0.036825
Note: FMLC:1. field-mixed	laboratory-com	pacted; LM	LC: laborate	ory-mixe	d laborat	ory compacted	; 2. DG: dense-	graded; GG:ga	p-graded.	

 Table A.2: Summary of Shear Laboratory Test Results for RAC Mixes (Temperature Effect).

(FMLC, AV =  $6.0 \pm 0.5\%$ , AC=8%).

Specimen ID	Binder Type	Comp.	Grad.	AV (%)	AC (%)	Test Stress Level (kPa)	Test Temp. (°C)	G (MPa)	n@5% PSS	PSS@ 5000 cycles
MAC15-6-13-3-7045	MAC15	FMLC	GG	5.5	7.4	70	45	162	40,070,646	0.011830
MAC15-6-1-3-7045	MAC15	FMLC	GG	5.7	7.4	70	45	176	19,114,642	0.009073
MAC15-6-15-3-7045	MAC15	FMLC	GG	6.4	7.4	70	45	175	6,089,300	0.014695
MAC15-6-9-3-7045	MAC15	FMLC	GG	6.5	7.4	70	45	197	7,304,479	0.012701
MAC15-6-10-1-7055	MAC15	FMLC	GG	5.9	7.4	70	55	70	282,137	0.022781
MAC15-6-10-3-7055	MAC15	FMLC	GG	5.8	7.4	70	55	69	341,817	0.023404
MAC15-6-11-2-7055	MAC15	FMLC	GG	6.0	7.4	70	55	49	74,117	0.033185
MAC15-6-11-3-10045	MAC15	FMLC	GG	6.0	7.4	100	45	77	44,152	0.033211
MAC15-6-15-2-10045	MAC15	FMLC	GG	5.9	7.4	100	45	55	217,422	0.027160
MAC15-6-9-1-10045	MAC15	FMLC	GG	5.7	7.4	100	45	178	1,807,288	0.016438
MAC15-6-2-3-10055	MAC15	FMLC	GG	6.3	7.4	100	55	48	1,546	0.070742
MAC15-6-6-3-10055	MAC15	FMLC	GG	6.3	7.4	100	55	72	16,274	0.040413
MAC15-6-7-2-10055	MAC15	FMLC	GG	6.2	7.4	100	55	66	9,225	0.045680
MAC15-6-11-1-13045	MAC15	FMLC	GG	6.5	7.4	130	45	70	2,048	0.063363
MAC15-6-8-1-13045	MAC15	FMLC	GG	6.2	7.4	130	45	157	385,830	0.021900
MAC15-6-8-3-13045	MAC15	FMLC	GG	5.9	7.4	130	45	144	187,093	0.025595
MAC15-6-1-3-13055	MAC15	FMLC	GG	5.7	7.4	130	55	41	447	0.077662
MAC15-6-2-1-13055	MAC15	FMLC	GG	6.5	7.4	130	55	46	428	0.058382
MAC15-6-7-3-13055	MAC15	FMLC	GG	6.0	7.4	130	55	76	11,883	0.041680
Note: 1. FMLC: field-mixed labo	oratory-compaci	ted: LMLC:	laboratory	-mixed la	boratory	compacted.				

Table A.3: Summary of Shear Laboratory Test Results for MAC15 Mixes (Temperature Effect).

(FMLC, AV =  $6.0 \pm 0.5\%$ , AC=7.4%).

ry. ry compa

2. DG: dense-graded; GG:gap-graded.

Specimen ID	Binder Type	Comp.	Grad.	AV (%)	AC (%)	Test Stress Level (kPa)	Test Temp. (°C)	G (MPa)	n@5% PSS	PSS@ 5000 cycles
MB15-6-6-1-7045	MB15	FMLC	GG	6.9	7.1	70	45	80	34,680	0.034050
MB15-6-6-3-7045	MB15	FMLC	GG	6.3	7.1	70	45	74	23,359	0.037937
MB15-6-9-1-7045	MB15	FMLC	GG	6.1	7.1	70	45	78	18,166	0.035066
MB15-6-2-2-7055	MB15	FMLC	GG	6.5	7.1	70	55	59	7,560	0.039938
MB15-6-10-3-7055	MB15	FMLC	GG	5.9	7.1	70	55	50	3,128	0.056893
MB15-6-2-3-7055	MB15	FMLC	GG	6.5	7.1	70	55	22	492,011	0.028120
MB15-6-3-1-7055	MB15	FMLC	GG	6.0	7.1	70	55	51	85,441	0.026907
MB15-6-4-3-7055	MB15	FMLC	GG	5.8	7.1	70	55	37	320,841	0.014645
MB15-6-5-2-7055	MB15	FMLC	GG	6.4	7.1	70	55	52	7,912	0.043635
MB15-6-7-1-7055	MB15	FMLC	GG	6.5	7.1	70	55	56	8,531,076	0.016734
MB15-6-10-1-10045	MB15	FMLC	GG	6.3	7.1	100	45	78	2,630	0.062278
MB15-6-11-1-10045	MB15	FMLC	GG	5.8	7.1	100	45	69	48,279	0.031921
MB15-6-9-3-10045	MB15	FMLC	GG	5.7	7.1	100	45	80	3,265	0.059291
MB15-6-10-2-10055	MB15	FMLC	GG	6.3	7.1	100	55	53	895	0.085694
MB15-6-11-3-10055	MB15	FMLC	GG	5.7	7.1	100	55	43	3,276	0.056567
MB15-6-8-3-10055	MB15	FMLC	GG	6.0	7.1	100	55	44	1,189	0.064759
MB15-6-2-1-13045	MB15	FMLC	GG	6.2	7.1	130	45	81	859	0.086344
MB15-6-6-2-13045	MB15	FMLC	GG	6.5	7.1	130	45	80	1,108	0.086964
MB15-6-8-2-13045	MB15	FMLC	GG	5.5	7.1	130	45	84	2,351	0.068717
MB15-6-11-2-13055	MB15	FMLC	GG	5.8	7.1	130	55	60	1,239	0.070906
MB15-6-5-3-13055	MB15	FMLC	GG	5.9	7.1	130	55	39	223	0.088387
MB15-6-7-2-13055	MB15	FMLC	GG	5.5	7.1	130	55	54	1,415	0.061114
Note:										

Table A.4: Summary of Shear Laboratory Test Results for MB15 Mixes (Temperature Effect).

(FMLC, AV =  $6.0 \pm 0.5\%$ , AC=7.1%).

Note:

1. FMLC: field-mixed laboratory-compacted; LMLC: laboratory-mixed laboratory compacted.

2. DG: dense-graded; GG:gap-graded.

Specimen ID	Binder Type	Comp.	Grad.	AV (%)	AC (%)	Test Stress Level (kPa)	Test Temp. (°C)	G (MPa)	n@5% PSS	PSS@ 5000cycles
MB4-6-1-4-7045	MB4	FMLC	GG	5.8	7.2	70	45	98	503,010	0.017837
MB4-6-5-3-7045	MB4	FMLC	GG	6.4	7.2	70	45	85	1,238,930	0.017178
MB4-6-7-2-7045	MB4	FMLC	GG	5.8	7.2	70	45	86	518,147	0.019745
MB4-6-15-1-7055	MB4	FMLC	GG	5.7	7.2	70	55	58	27,623	0.033518
MB4-6-16-1-7055	MB4	FMLC	GG	5.5	7.2	70	55	55	21,487	0.035499
MB4-6-2-1-7055	MB4	FMLC	GG	6.5	7.2	70	55	59	48,738	0.030014
MB4-6-1-2-10045	MB4	FMLC	GG	6.3	7.2	100	45	100	129,823	0.024905
MB4-6-2-3-10045	MB4	FMLC	GG	6.3	7.2	100	45	83	34,513	0.031734
MB4-6-5-1-10045	MB4	FMLC	GG	5.8	7.2	100	45	94	14,843	0.036867
MB4-6-12-1-10055	MB4	FMLC	GG	6.1	7.2	100	55	51	4,532	0.051366
MB4-6-12-3-10055	MB4	FMLC	GG	6.1	7.2	100	55	47	4,348	0.045655
MB4-6-7-3-10055	MB4	FMLC	GG	6.3	7.2	100	55	56	1,820	0.023518
MB4-6-3-3-13045	MB4	FMLC	GG	5.8	7.2	130	45	85	3,107	0.058447
MB4-6-4-2-13045	MB4	FMLC	GG	5.5	7.2	130	45	73	6,539	0.047531
MB4-6-7-1-13045	MB4	FMLC	GG	6.4	7.2	130	45	86	9,400	0.041901
MB4-6-3-1-13055	MB4	FMLC	GG	6.5	7.2	130	55	48	276	0.078451
MB4-6-3-2-13055	MB4	FMLC	GG	5.5	7.2	130	55	45	588	0.062037
MB4-6-5-2-13055	MB4	FMLC	GG	5.9	7.2	130	55	47	556	0.077945
Note:	•	•		•	•	•	•	•		

Table A.5: Summary of Shear Laboratory Test Results for MB4 Mixes (Temperature Effect).

(FMLC, AV =  $6.0 \pm 0.5\%$ , AC=7.2%).

Note:

FMLC: field-mixed laboratory-compacted; LMLC: laboratory-mixed laboratory compacted.
 DG: dense-graded; GG:gap-graded.

Specimen ID	Binder Type	Comp.	Grad.	AV (%)	AC (%)	Test Stress Level (kPa)	Test Temp. (°C)	G (MPa)	n@5% PSS	PSS@ 5000 cycles
DGAC-69-3-2-7045	AR4000	FMLC	DG	9.4	5.0	70	45	252	2,186,262	0.011442
DGAC-69-3-3-7045	AR4000	FMLC	DG	9.5	5.0	70	45	220	5,929,144	0.011293
DGAC-69-1-1-7055	AR4000	FMLC	DG	9.0	5.0	70	55	76	25,459	0.030580
DGAC-69-1-3-7055	AR4000	FMLC	DG	9.5	5.0	70	55	72	4,131	0.054568
RACG-69-1-2-7045	ARB	FMLC	GG	9.2	8.0	70	45	169	189,749	0.023460
RACG-69-2-2-7045	ARB	FMLC	GG	8.8	8.0	70	45	158	16,341	0.037432
RACG-6-1-3-7055	ARB	FMLC	GG	9.4	8.0	70	55	70	18,987	0.038197
RACG-69-1-1-7055	ARB	FMLC	GG	9.2	8.0	70	55	77	27,352	0.034081
RACG-69-2-3-7055	ARB	FMLC	GG	8.8	8.0	70	55	73	13,055	0.042373
MAC15-9-3-2-7045	MAC15	FMLC	GG	8.6	7.4	70	45	147	19,981	0.035773
MAC15-9-8-2-7045	MAC15	FMLC	GG	8.6	7.4	70	45	115	9,008	0.046281
MAC15-9-4-1-7055	MAC15	FMLC	GG	9.1	7.4	70	55	63	7,133	0.047294
MAC15-9-7-1-7055	MAC15	FMLC	GG	8.5	7.4	70	55	64	5,195	0.040399
MB15-9-15-1-7045	MB15	FMLC	GG	9.2	7.1	70	45	73	17,708	0.038993
MB15-9-15-2-7045	MB15	FMLC	GG	9.0	7.1	70	45	79	14,934	0.038910
MB15-9-13-2-7055	MB15	FMLC	GG	8.5	7.1	70	55	47	1,475	0.049630
MB15-9-13-3-7055	MB15	FMLC	GG	8.7	7.1	70	55	47	2,717	0.056306
MB15-9-14-2-7055	MB15	FMLC	GG	9.2	7.1	70	55	138	4,238	0.028272
MB4-9-19-2-7045	MB4	FMLC	GG	9.1	7.2	70	45	78	527,488	0.019540
MB4-9-20-1-7045	MB4	FMLC	GG	9.1	7.2	70	45	64	14,370	0.039149
MB4-69-1-3-7055	MB4	FMLC	GG	9.1	7.2	70	55	72	4,131	0.054568
MB4-9-17-1-7055	MB4	FMLC	GG	8.6	7.2	70	55	46	3,609	0.053601
MB4-9-18-3-7055	MB4	FMLC	GG	9.1	7.2	70	55	54	3,145	0.056091
MB4-9-19-1-7055	MB4	FMLC	GG	8.6	7.2	70	55	60	8,259	0.043312
Note: 1. FMLC: field-mixed labo	ratory-compact	ed; LMLC:	laboratory-	mixed lat	oratory	compacted. 2.I	OG: dense-grade	ed; GG:gap-gra	aded	

 Table A.6: Summary of Shear Laboratory Test Results for all Mixes (Air-Void Content Effect)

(FMLC,  $AV = 9 \pm 1\%$ ).

Specimen ID	Binder Type	Comp.	Grad.	AV (%)	AC (%)	Test Stress Level (kPa)	Test Temp. (°C)	G (MPa)	n@5% PSS	PSS@ 5000 cycles
DGAC-6-21-1-LT-7045	AR4000	FMLC	DG	5.9	5.0	70	45	480	12,957,636	0.010262
DGAC-6-21-3-LT-7045	AR4000	FMLC	DG	6.1	5.0	70	45	480	22,551,627	0.007604
DGAC-6-18-3-LT-7055	AR4000	FMLC	DG	6.3	5.0	70	55	191	538,053	0.016410
DGAC-6-19-1-LT-7055	AR4000	FMLC	DG	5.5	5.0	70	55	223	715,540	0.014614
RACG-6-8-2-LT-7045	ARB	FMLC	GG	6.4	8.0	70	45	102	897	0.049862
RACG-6-13-1-LT-7045	ARB	FMLC	GG	5.9	8.0	70	45	340	924,374	0.013866
RACG-6-13-2-LT-7055	ARB	FMLC	GG	6.5	8.0	70	55	156	116,446	0.027573
RACG-6-16-3-LT-7055	ARB	FMLC	GG	6.1	8.0	70	55	245	44,248	0.028600
MAC15-6-1-2-LT-7045	MAC15	FMLC	GG	5.8	7.4	70	45	91	16,526	0.038512
MAC15-6-2-1-LT-7045	MAC15	FMLC	GG	6.5	7.4	70	45	172	72,628	0.028884
MAC15-6-5-1-LT-7055	MAC15	FMLC	GG	5.8	7.4	70	55	119	14,708	0.038381
MAC15-6-5-2-LT-7055	MAC15	FMLC	GG	6.1	7.4	70	55	98	14,152	0.040559
MB15-6-12-2-LT-7045	MB15	FMLC	GG	6.0	7.1	70	45	90	315,458	0.017876
MB15-6-12-3-LT-7045	MB15	FMLC	GG	5.9	7.1	70	45	91	180,700	0.027282
MB15-6-12-1-LT-7055	MB15	FMLC	GG	5.7	7.1	70	55	63	54,675	0.029856
MB15-6-2-2-LT-7055	MB15	FMLC	GG	6.5	7.1	70	55	59	7,560	0.039938
MB4-6-10-1-LT-7045	MB4	FMLC	GG	5.5	7.2	70	45	100	245,601	0.021296
MB4-6-9-3-LT-7045	MB4	FMLC	GG	6.3	7.2	70	45	100	781,924	0.019624
MB4-6-10-2-LT-7055	MB4	FMLC	GG	5.6	7.2	70	55	68	283,823	0.025628
MB4-6-10-3-LT-7055	MB4	FMLC	GG	6.5	7.2	70	55	71	156,258	0.025026

Table A.7: Summary of Shear Laboratory Test Results for all Mixes (Aging Effect).

(FMLC, AV =  $6 \pm 0.5\%$ ).

Note:

FMLC: field-mixed laboratory-compacted; LMLC: laboratory-mixed laboratory compacted.
 DG: dense-graded; GG:gap-graded.

Specimen ID	Binder Type	Comp.	Grad.	AV (%)	AC (%)	Test Stress Level (kPa)	Test Temp. (°C)	G (MPa)	n@5% PSS	PSS@ 5000 cycles
DGAC-FIELD-01-TL-7045	AR4000	FMFC	DG	5.6	5.0	70	45	161	1,732,691	0.018420
DGAC-FIELD-02-TL-7045	AR4000	FMFC	DG	6.7	5.0	70	45	214	125,081	0.021524
DGAC-FIELD-05-TL-7045	AR4000	FMFC	DG	6.3	5.0	70	45	193	2,788	0.060184
DGAC-LM-6-2-1-7045	AR4000	LMLC	DG	6.2	5.0	70	45	98	8,116	0.042107
DGAC-LM-6-2-2-7045	AR4000	LMLC	DG	6.1	5.0	70	45	97	7,872	0.044650
DGAC-LM-6-2-3-7045	AR4000	LMLC	DG	6.3	5.0	70	45	104	36,311	0.032639
RAC-LM-6-1-1-7045	ARB	LMLC	GG	5.7	8.0	70	45	143	31,352	0.036181
RAC-LM-6-1-2-7045	ARB	LMLC	GG	5.7	8.0	70	45	139	46,515	0.032539
RAC-LM-6-2-3-7045	ARB	LMLC	GG	5.7	8.0	70	45	209	646,689	0.016811
MAC15-LM-6-1-1-7045	MAC15	LMLC	GG	6.1	7.4	70	45	77	1,183,841	0.022451
MAC15-LM-6-1-2-7045	MAC15	LMLC	GG	6.0	7.4	70	45	81	611,185	0.016737
MAC15-LM-6-1-3-7045	MAC15	LMLC	GG	5.6	7.4	70	45	90	51,750,300	0.015055
MAC15-LM-6-3-2-7045	MAC15	LMLC	GG	5.9	7.4	70	45	93	616,304	0.024981
MB15-LM-6-2-2-7045	MB15	LMLC	GG	6.2	7.1	70	45	34	10,236	0.042000
MB15-LM-6-2-3-7045	MB15	LMLC	GG	6.2	7.1	70	45	51	21,105	0.033786
MB4-FIELD-24-TL-7045	MB4	FMFC	GG	7.1	7.2	70	45	74	74,487	0.035622
MB4-FIELD-25-TL-7045	MB4	FMFC	GG	6.3	7.2	70	45	68	70,284	0.033886
MB4-FIELD-26-TL-7045	MB4	FMFC	GG	6.1	7.2	70	45	60	303,840	0.032545
MB4-LM-6-1-2-7045	MB4	LMLC	GG	5.5	7.2	70	45	84	39,148	0.034497
MB4-LM-6-1-3-7045	MB4	LMLC	GG	5.7	7.2	70	45	74	283,016	0.026028
MB4-LM-6-2-3-7045	MB4	LMLC	GG	5.7	7.2	70	45	82	1,696,084	0.022477

 Table A.8: Summary of Shear Laboratory Test Results for all Mixes (Compaction Effect).

(FMLC&LMLC, AV =  $6 \pm 0.5\%$ ).

Note:

1. FMLC: field-mixed laboratory-compacted; LMLC: laboratory-mixed laboratory compacted.

2. DG: dense-graded; GG:gap-graded.

3. All FMFC specimens are in Field Aged and Trafficked condition.

			(FML	C&LML	C, AV	$= 6 \pm 0.5\%$ ).				
Specimen ID	Binder Type	Comp.	Grad.	AV (%)	AC (%)	Test Stress Level (kPa)	Test Temp. (°C)	G (MPa)	n@5% PSS	PSS@ 5000 cycles
MB4-DG-6-1-3-7045	MB4	LMLC	DG	5.6	6.3	70	45	91	19,407	0.028693
MB4-DG-6-3-1-7045	MB4	LMLC	DG	6.3	6.3	70	45	81	31,889	0.029908
MB4-DG-6-3-3-7045	MB4	LMLC	DG	6.2	6.3	70	45	91	100,053	0.021910
MB15-DG-6-1-1-7045	MB15	LMLC	DG	6.1	6.0	70	45	78	326,231	0.018156
MB15-DG-6-1-3-7045	MB15	LMLC	DG	6.3	6.0	70	45	91	43,389	0.025670
MB15-DG-6-2-1-7045	MB15	LMLC	DG	6.2	6.0	70	45	72	22,003	0.031358
MAC15-DG-6-1-1-7045	MAC15	LMLC	DG	6.2	6.0	70	45	162	6,554,338	0.012057
MAC15-DG-6-2-1-7045	MAC15	LMLC	DG	6.5	6.0	70	45	240	3,658,443	0.011243
MAC15-DG-6-2-3-7045	MAC15	LMLC	DG	6.3	6.0	70	45	207	2,983,215	0.013458
Note:										

 Table A.9: Summary of Shear Laboratory Test Results for all Mixes (Gradation Effect)

FMLC: field-mixed laboratory-compacted; LMLC: laboratory-mixed laboratory compacted.
 DG: dense-graded; GG:gap-graded.

# **APPENDIX B: MIX DESIGN INFORMATION**

DGAC

	A TRET IN		
	XIN 0 6 2002		To TRY when potent
30	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		STATE OF CASTOR AND
<u>د</u>	10072-101		The second start Same Same
i.	The section of the se		
*	REPURT OF TESTS ON		Burner in Burner wer 10002-10
	AO Arian B. C. I	A THE ALL AND A LAND A THE ALL AND A	I all and the second second the second
	AC ASCESSATE (PUDICE LANPLE)	X 117 + 4. X 101 (PUL)	CEEDENALS INFO
	IF CONTRACT, BIT CANTRACT ITEM	Tot I am HALL	DUPTILAS WITH A DO
	Meg HLB Carlaba And Indertain	He I STAR ISCHINGT	() CHAND IND
		Way to the Interior	1) BASCINI TEETA
	THE PRESS AND	X INT . Y Y	SAMPLE OF _ DUENDED EAMPLE
E			TOR BUE INALL ALL ALL ALL ALL ALL ALL ALL ALL ALL
5	AUTO ANT	BURTURE VAPOR SUSCEPT	MUX KEG A- Them Source April
2	Z PER S	WIT NIME DAT	CVAIL EL-
10	1% Patrate	1/4/-100 1120 39 30	NPIN
		14. #11-8	LICATION OF SOURCE FLAM A I
ö		- 94. 0712	>MK Lipi Hanmar.
<del></del>	5 14 60-73	APRELEIC OPAVIAN ACCOMUNAT	T SANTER M. A ABSS TO FRANKER
AX	8 - 10	1 NUCTO + 1.83 6.7 44	BROUGHER SYAR Like How -
<u>ر</u>	19 19 10 10 10	ABI TAULAD 7 C-	Splat southing ( ) Tess atoress Maping
		ABRASION TESTS Y LAPA PILOT.	IEAALS
		LAS ANK(10-1001 _ 1/2 D	-AAST 491-48-5002
	× 11 × 12	10 ANELLO-3000 _1 10 45	
	107 6 50 3.61	FILM STREPANG	- AL FOR ASSIGN
	220 4 5-7 10 1.4	CALISKED PARTICLES	PATE SAMPARA 514/02
3		") "BAX " [ ! ] 4	17 Mi Dand Inc RC
끸	AVASONID A TANK TO TAKE TO TAK	ANA E HOW CATER YOLDE : 12 STICH. FO	0131 00. ATT. PA. 04 - SUL-37-12.7/150.
912	TTYPHUMME - Gold C	Training a server tartes 7.9 - 3.2	V. F. D. T. S. T.
2	Leditre	Chedinia & rear Lis Ab Tostan	UNIN RELATION OF ME SELENCE State
z	en aus AR ODO	11. 10. 1137 ME 1210-2300	04-0T 14 H
<b>a</b> :	17. ANTO 45 5.0 5.5 6.0	Q BOT2-9C BW 41	NEP. 10. ACAH- FO3210461E
1	12-32 10-33 14-36 13-3F 1		RES. ENGA, OR SUPP. LINE BALLES
- 04	37 48 49 42 42	H+++	ATIONESS TRUNCH
	YORU		CONTRACTOR OL JONG
নি	R6 7.1 6.3 4.8		a series a series and a series and a series and a series of the series o
2	D The IA AS	PELLAR:	
		18 9 HEN O FLUSMED BY HE	
		MODELOTELY YVPILL BING	
5	48 - 34	LETTOYUSED-	CATHRINA 8. \$14305
	400 . 6		DISTRICT RALLER STRUCT
Ś			SHANCHER HARACER
53	· · · · · · · · · · · · · · · · · · ·		A CONTRACTOR OF A CONTRACTOR O

Figure B.1: Caltrans DGAC mix design.

RAC-G

ALST NO. SATE ARCTI ALS TY THIS		man - rel main - · · ·
· · · · · · · · · · · · · · · · · · ·	D. DISTRICT COMMENTS IN LAND	E
Cate Anna	I and a state of the state of t	71-101 (AP 27) 0ANPLE 10 1-4
ADD's GOA	A BULL BULL BUILT DEMODENT	DEPARTMENT OF TRANSPORTATION COOD
10063-61C. " soft woores A. A. H. 644	ALI SEPTEMENTALISTES	TIPALININARYTERIE SANAL SUNTE 423512
REPORT OF TESTS ON	C construction acor. (1)	C. PSECTOR INTE "INQUIRE LAR OFFICE
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	TELENART FAILER	Bist for the
LA STATE AND STATE AND		LI ALCOMARCETISTE LI BRANCH LAB HO 10063 61C
19mm. Max TYNE Q. Kubherned.	2 Minte	INEGASHODST DEST. LAD
IP CONTRACT OUN CONTRACT ITINE	Parlan state - State 1 sife	A L'ANTE LAN SIGNAGHI CO AN
moyets	F Ind A share water and the state of	HO HKC. NO
		AUTHORIZATION
HAR PARA APPLICADAST - LON COLUTER AND UNT		Dabreis Ha
	I WE LY MAN IN THE POINT	I SAMPLE OF 14.00 ML AC. MIX BLENded Sample
M MIL ADDAD WERD'S ET MATAGE MET	NOILYUNE VARDALLUSCERT	POR USE IN YPE B Rubberized.
ALAN THE STORE STORE ST. VIC LANTI . CALTARY AND	AND TO BE AND THE PARTY AND THE	
12 4	healar. Ad som p	ANDLY FACH SYAR LAKE HEAMAN AL.
- I Will all all all all all all all all all	TINIL CATE	
- 100 - 100 will sewin	hlinung	DEAM
	1 th W. C.H.	CUCATION OF BOURCE PYATL CALOP HERLIAN GUNA
7 20-86	No.4 WE No=	
Find Gtory	IPECIFIC CHAVITY ADSHEGATE	is tally to an in the transfer to the transfer to the tally the tally to the tally
A T/ 2.3	14 ASSISTOF F 7 579 6-3,60	SWARE OF SURA TALL THE ME ALL ME
	PTREASHING P	AND ALL STATE STAR. TAL
15 12	ABRASION TESTS CALD ALLAY	PALAMA
30 10	C03 44 66 144 - 100	ACHARACTER DECEMBER OF
30 3	LOA ANGLES - PEDE - 1. P. MOL	- OII ABCOMMENDALION.
100 5	and the second	and summer, many second an angel - many
200 31 0 - 7 14	PILATITRE PINO SA -TY	
Istyled p	LINUSHED PARTICIPES	DATE SANTAD_3-6-03
MIA.	AU & 100 AGE 1-14 (92-16 TOUR.	IT'S STAWART THE last Lat Tal
FREENEN A A	AND TOR MATERS AND ANT AND ANT AND	All a min Alt SAL DO ADIT
ICHAPTERTYPE WITH TEM STATE	TEPH LEPT (R Terupats' Cart) per	the the mill the france out of a faith
Comp Life dater	Sentervelt tanders at settings	litu.yt
NIT. SALLE RUASPRITES THAT CHAT	The second states of the second second	
1. AALIO 7.0 7.5 8.D. 25 18.0	PF Her. Prov Ban / Blace 17/1 ga	CONT IN OCHOTOS OLL
17 40. Die 2.232 2. 247 2.210 0.253	AST PILE-STABALE!	
PECIP. STADIOMETER	A AND ANTZ	HE WER ON WOR ANDRENI NAMERICI
2300 28 31 28 26 1.30	100/1 1043-514 D/WF3	ADDATSS
VOIDE	PPA- Gacusey File	CONTRACTOR D. C TONEC.
7.8 6.4 5.4 45121		Juli Cut
SPECIF VMA		MALL TO CAME DEPENDENCE
BANIE IS ISTISTIC. TR	A MAN	SAME DESTINATION AS SAMPLE
	1 2.0%. YOIOS - H. 8-02C	1 3-780
X JAU'ES	Accommented Range: 7.9	-B.L KAMPS
1/2- 83 58-19	V CONTRACTOR	
18: 69 134-9	D. X045 - 8.8- 080	DELECTION DELECTION
IF4 - 34	THE REAL REAL B.S.	B.B.
take (LOD) has write		ENGRONING, BULKALSA
		Him, see not son a sur

Figure B.2: Caltrans RAC-G mix design.

TEST R	ev. 5/9	BT B)						•••					BTATE OF	TRANSPORTATIO	° <b>C</b> 23	6791
TEST NUME	EA		DATE	RECEIVED	•	_	Dt	STRICT	ENGINEER	ME	TS		PRELIMINARY TESTE	SAMPLE SENT I	0	
			CAL	ne 10	200	3	Df	ST. MAT	LS. ENGINEER	PA	VEMENT		PROCESS TESTS	THOOTHS. LAB	FIELD NO	1100
07	23	028	G	C		nh.	RE	ESIDENT	ENGINEER	AC	COUNTIN	G	ACCEPTANCE TESTS	BRANCH LAB	LABINOS	1120
			DAT	REPORTED			co	ONSTRU	CTION DEPT.	$X_3$	09 F	lice	INDEPENDENT	DIST. LAB		
		050007	10	une 1	3, 20	03	31	0 E	T & GRADE		SP. GR.		ASSURANCE TESTS		LOT NO	
		HEPURI	UFI	:515 UN	~~~~		21	7 S.E.		30	SWELL		DIST. LAB	NO	_ REQ. NO	
1 mi	m	ype	6	Mix	Des	134	30	2 FILM	STAIP	.31	I MOIST.		TRANS. LAB	AUTHORIZATION		
- 1							30	4.1 STA	B. (CONTROL)	300	O CK.E		SPECIAL TESTS	NO		
WITL	<u>م</u>	<u>m B -</u>	Ч				$\times$	4 2 STA	B. (DESIGN)	X*	7 080		SAMPLE OF AC A	sire, + 1	MB-4	Bude
-		IF CONTRACT	USE CO	NTRACTITE	м	•	30	7 M.V.S		X3	66 5	TAD	FOR USE IN 19 MM	Type G	MIX DI	Sic H
DIST i		CHG.	AUTHOR		SUB	JOB		MOIS	TURE VAP	ORSI	USCEP	T	with Vales	0 M 13-4	Binde	× ~
5,93	20	519 :	9111	16181		1 1	HOUR	S				SPECIF.	SAMPLE FROM ALSE	SATE - S	var Lu	Ke the
SPEC	IAL DES	IGNATION	FA	CTIVITY	AMOU	NT	MOIST	. ABSC	RB.				Hermin Bi	nder - Vi	ler il	
(ประ	WINSO A	ppwcable)	1.0	(OBJECT			STABI	LOMET	ER				DEPTH			
<u> </u>	AS	AET.	AS USED	SPECIF	SURFACE	SURFACE	BIT, A/	OITA					LOCATION OF SOURCE			
REC	CEIVED	CRUSHED BY	VOL. BY	n Li tuna	AREA I FACTORS	AREA	SP. GF	A. BRIC								
50 mm					NOTE:		K.,	TOIL	KI.	K	m-		THIS SAMPLE 43	AND IS ONE OF	SAMPLES	
375mm					All Se Area F	actors	51	ECIP	IC GHAVII	YAG	GREG	ALE	(NO. CONTAIRERS)	_A GEOUP OF	(YOKS, GALS., BALS.	, STA., ETC.)
150mm ]0	0				Must 8	Used	ASRE	CEIVE	DF 27	1 C	2.6	0	NANUFACTURER CAL	TRANS		
190 mm	0				in Calc	dations	RET. C	CRUSH	EDF	- C			TOTAL QUANTIN IN A	TEST RESULTS WORK	AL	1-13-11
	2						ABHI	ASIU	TESIS		LUSS	SPECIF.	REMARKE ( T	NY CT	305 (3	300
615 000							LOSA	NGELE	S - FOODay				CT 3610	1 - 31.7	(T39	32
475mm	21					. 0.4	1007									
2 15 mm					0.4		FILM	STR	PPING							
1 18 mm	13				16			C	RUSHED P	ARTIC	CI ES		COVER ADDITIONAL INFORMATION	WITH LETTER	~	
910 µm	U				2.9		(C)		(F)		COMB.		DATE SAMPLED	në S, Ju	ne	200 3
300 µm	.7				6.1		SANDE	OUIVAI	FUT VALUE -		SPECIE		BY Cornell	TITLE	mRIL	17
150 µm	5				12.3		RECON	MENDE	DBITUMENCO	NTENT	7	2	DIST., CO., RTE., P.M			
75 µm	3.1				32.8		0	SHADIN	IG AS USED V	WASOB	TAINED	BY				
BITUMEN					SURFACI	AREA		COMB	NING SAMPL	ES AS I	FOLLOW	<u>s</u>	LIMITS			
		<u> </u>			1		WT.	VOL	7ES7 NO.		DESCRIP	TION			· · · · · ·	
SPECIM		A	8	C		E	25%		0063-5-	2dB	in #		CONT. NO. 59-9	2650		
ICAIPEHAI	UNE	mix	Lem	PP	165	6	55%	,	10063-59	IC B	in V	2	FED. NO			
BIT. GRADE	E	man	BL	m R-LI	MALL	A.L.	30%		0063-59	dB	in #	3	RES. ENGR. OR SUPT	Mike (	-ccK	
BIT. RATIO		7.0	7.5	8.12	0.5	5.0	20%		10063-60	04 E	Sin B	4	ADDRESS			
SP. GR. BR	D.	2.304 2	344	2 330	2369	7.201							CONTRACTOR			
SPECIF.		STA	BILOM	TER			1			_					1	
21 m	in	331	31	32	30	32	1						MAIL TO SAME	DESTINATION	N AS SAM	IPLE
			VOIDS					DVe		_			· · · · · · · · · · · · · · · · · · ·	<u> </u>		
COLOIS	· ·	4.79	433	230	0.08	0.17	1.)	H. P	1/2 Var	15	37	2	ABC CT	382 CF	& 482°C	- :
SPECIF.			SHEE	Vm	Α		. · ·	0		S.	7.	1	~~ ~ ~ ~ ~		1 427	
10 M	IN	15 1	14	15	14	15	C	Rec	-ommen	n <i>oev</i>	R.	en e	10-112		11-13 4	
		SURF	CE FLL	SHING			10r	AU	alun	0	458-0	A 1	1 D-4 # 1006	John C.		

Figure B.3: Caltrans MB-4 mix design.
STATE CALIFORNIA • DEPARTMENT OF TES ORT TL-0302 (Rev. 5/96)	TRANSPORTATION		TL-101 (REV. 8-76) BTATE OF CALIFORNIA DEPARTMENT OF TRANSFORTATION
TEST NUMBER DATE RECE	IVED 2003	DISTRICT ENGINEER METS	PRELIMINARY TESTS SAMPLE SENT TO -235/92
0x 3 029	APPROV.	RESIDENT ENGINEER ACCOUNTING CONSTRUCTION DEPT. 309 R.L.	ACCEPTANCE TESTS BRANCH LAB DET LAB NO.
REPORT OF TESTS	13,2003	310 202.3 EXT & GRADE Store SP. GR. 217 S.E. Store Sto	ASSURANCE TESTS LOT NO DIST. LAB SHIPMENT P. O. OR
igmini type G	mix	302 FILM STRIP .311 MOIST.	TRANS, LAB
Design with ME	RAIE	104.1 STAB. (CONTROL) 303 CKE	SPECIAL TESTS NO
IF CONTRACT, USE CONTRACT		307 M.V.S. VI21 L CT.A.h	SAMPLE OF 19mm 14c Aug + MB+5 BING.
SOURCE CHG. EXPENDITURE	SUB-JOB	MOISTURE VAPOR SUSCEPT	FOR USE IN 19 MM TYPE G. MILK Design
5:4 3:214 5:9: 1911214	Q10 1 1 1 1	HOURS SPECIF.	With MB-15 Binder from Visler U
SPECIAL DESIGNATION	Y AMOUNT	MOIST, ABSORB.	VALUND YAR MAKE HERMAN +
		STABILOMETER	
AS RET. AS USED	SPECIF. SURFACE SURFACE	SP GR BBO	LOCATION OF SOURCE SUME LAKO HECKING
10 mm	SOUGHT FACTORS AREA	Kee Kie Kma	Valero
375 mm	All Surface	SPECIFIC GRAVITY AGGREGATE	THIS SANPLE HA AND IS ONE OF SANPLES
259 mm 100	Area Factors Must Be Used	AS RECEIVED F 2.79 C 2.60	(NO. CONTAINE PE) (TONE, GALS., BELE., STA., ETC.)
190 mm 68	In Calculations	RET. CRUSHED F C	MANUFACTURER C. ALTRANS
125 mm 82		ABRASION TESTS % LOSS SPECIF.	AVAILABLE TAMP & TEST RESULTS HORMAL AVAILABLE TOPSIRED HORMAL
615 mm -		LOS ANGELES - 100Rev	REMARKS CT 309, CT 309, CT 309
415mm 76	0.4	LOS ANGELES - SOURGY	CT 366, CT 367, CT 392
215mm 2.1	0.4	FILM STRIPPING	
1.19 mm   3	1.6	CRUSHED PARTICLES	COVER ADDITIONAL INFORMATION WITH LETTER
	2.9	(C) (F) COMB.	DATE SAMPLED JUNE 5, JUNE 9, 2003
sioum 5	6.1	SAND EQUIVALENT VALUE = SPECIF. =	BY LEPOSE COMALL TITLE MURE H
/5 um 3, ]	32.8 SUBFACE AREA	GRADING AS USED WAS OBTAINED BY COMBINING SAMPLES AS FOLLOWS	DIST., CO., RTE., P.M
SPECIMEN A 1 D	m <sup>2</sup> /kg	WT. VOL. TEST NO. DESCRIPTION	
TEMPERATURE DOUVITE	LITOL	25% 10063-576 Bin #1	CONT. NO. 51-712 680
Comportion To	P 410 0 154"/	25% 10063-58 C Bin #2	FED. NO
BIT. GRADE MB-15 MB-15 MB-1.	5 mB-15 mB-15	30% 10063-596 Bin # 3	RES. ENGR. OR SUPT. SVIIK & Cerk
BIT. RATIO 7.0 7.5 8.0	8.5 9.0	2014 10063-60 C Bin 94	ADDRESS
SPECIF. STABIL CALETER	3 2.347 2.348		CONTRACTOR
23 min 36 33 29	30 30		MAIL TO SAME DESTINATION AS SAMPLE
4.46 2.62 2.3	8 0.97 0.51	REMARKS	
	nit	· · · · · · · · · · · · · · · · · ·	OBL CFE4B2C=1.2
Umin 15 15 15	15 15	Recommended Rung	2 7.0-7.1
SURFACE FLUSHING	<u> </u>	Gradution based on B.	4 7 10067-61 C

Figure B.4: Caltrans MB-15 mix design.

TEOL		1								DEPARTMENT OF	TRANSPORTATIC	<sup>№</sup> <b>C</b> 23679
IEST NUMBER		DATE	ACCEIVED	200	2	DISTR	RICT ENGINEER	METS		PRELIMINARY TESTS	SAMPLE SENT TO	, 20010
023	030	CALC.	ue uz	APPROV.		0191.	MAT'LS. ENGINEER	PAVENE	IT .	PROCESS TESTE	HOOTRS. LAB	FIELD NO
	040	G.	Ζ,		rift	RESIL	ENT ENGINEER	ACCOUN	DHD	ACCEPTANCE TESTS	BRANCH LAB	DG0 6 Marshall
		DATE	HEPOHIEL	3 201	2	CONS	TRUCTION DEPT.	X309	Are	INDEPENDENT	DIST. LAB	
	REPOR	T OF TE	STS ON	st and	13	202.3	EXT & GRADE	X 308 SP. 0	R	ASSURANCE TESTS		LOT NO.
19 mm	Type	G	MIL	Dec		2175	.E.	305 SWE	L	DIST. LAB	SHIPMENT	P. O. OR
	11-	-		Deal	5 1	302 FI	LM STRIP	311 MOIS	т.	TRANS. LAB		- REQ. NO
WITH	MAC	15				304.1	STAB. (CONTROL)	303 CK.E		SPECIAL TESTS	AUTHORIZATION	
	HE CONTRA					X 304.2	STAB. (DESIGN)	X 367 OBC		SAMPLE OF AC	Acacento	. Mar 15
SOURCE	I CHG.	EXPENDI	TUBE	M	•	307 14	.V.S.	X366	STHD	FOR USE IN 19 AL	TYAN C- V	My Doc
DIST UNIT	DIST.	AUTHORIZ	ATION	SUB	OB	MC	STURE VAP	OR SUSCI	PT	with Mile	ERIN	The Design
PI2:3.51	5 5 9	19112	6 3 6			HOURS			SPECIF.	SANPLE FROM SULL	LAK It	
Use Wher	Applicable)	F.A OR	OBJECT	AMOUN	r	MOIST. AL	SSORB,			Binder	- MANNIN T	right /
		!		1 1 1	1 1	STABILO	ALIER			DEPTH		
AS	RET.	ASUSED	SPECIF	SURFACE	WRFACE	CO CO D					Suche I will	No. 1.1.
10 mm	Chushep B	VOL BYWI		FACTORS	AREA	Sr. Gr. B				Pariamount	yar me	110 0 11 11 11
37.5 mm		-		All Sur		SPEC	IEIC GRAVIT	Km=	ATE	THIS SAMPLE 'H T	AND IS ONE OF 1 SI	AMPLES
250mm 100	1			Area Fa	1073	AS DECEN		AGGREG	AIE	IS SHIPPED IN (NO. CONTAINERS)	_A GROUP OFRI	CPRESENTING
190mm 98	1 1			In Calcula	Jsed	DET COU	SUED F	<u> </u>	60	MANUFACTURER C	LTIPHNS	
125mm 92	İ İ					ARDACI	ON TECTO	<u> </u>	LODEOUT	TOTAL QUANTITES	TEST RESULTS   NORKA	L 1 12 /
95mm 69						LOS ANGE	LES 100Rev	78 203	SPECIF.	AVAILABLE TITOTE	DESIRED OFFICIE	TY DATE NEEDED
6 35 mm					04	LOS ANGE	LES · 500Rev			CT 261	7 21 200	<u> </u>
475mm 36				0.4							1 10 17 0	- 32 -
2.35 mm 2)				0.8		FILM ST	RIPPING					
1.18mm [3				1.6			CRUSHED P	ARTICLES		COVER ADDITIONAL INFORMATION	WITH LETTER	
eco ym				2.9		(C)	(F)	COM		DATE SAMPLED JUA	e. 5, + Jun.	e 9,2003
150 -				6,1		SAND EOUT	ALENT VALUE =	SPECIE	-	BYCornell	TITLE	MICH
75 um 2 1			ļ	12.3		RECOMMEN	DED BITUMEN CON	TENT 7.4		DIST., CO., RIE., P.M.		
BITUMEN			t	32.8	DEA	GRA	NING AS USED W	AS OBTAINE	BY			
				m²/kg	REA	96	I I I I I I I I I I I I I I I I I I I	ASFULLO	vs	LIMITS		
SPECRAEN	A	8		0	<b>c</b>	WT. VOL	TEST NO.	DESCR	PTION			
TEMPERATURE	MIXT	-mnp	14	501-	<u> </u>	26%	10063-576	Bink		CONT. NO. 59-9	12680	•
HONOTOTE	Lompia	ction	Temo	0 15	•/-	2.5%	10063-58	+Bm #	2	FED. NO.		
BIT. GRADE	MBC ISI	HALLS W	We 15 .	MEC 15 W	W/L C	30%	10063-590	BINE	3	RES. ENGR. OR SUPT.	Mike Lo	νK
BIL HATIO	7.0	7.5 9	3.0	8.59	D	2070	10063-60 0	- Bin #	4	ADDRESS		
SPECIE	9.54)	13170	299		-					CONTRACTOR		
73	31	2 -	H									
NO DO LYO	26	Voins	5	28 3	2		1			MAIL TO SAME	DESTINATION	AS SAMPLE
	7.15	1012	411			REMARKS	1	1				JUS OANDEE
PECIF.		SWELL 1	121		-	(I.) H	0% 00	ids >	7.4	0B1 CT382	CF: 115	57 @ 492°C
19 m. n	171	161	(m th	-		Res		1 0	146-	51 - 511	-	1000
	SURF	ACE FI IISH	ING			1				11 - 14		
			14			Grac	ATION	DASPO	on Q	-4 # 100 62 -101 C	1	
the second second second second second second second second second second second second second second second se												

Figure B.5: Caltrans MAC 15 mix design.

Stage 4 Distribution Draft

## APPENDIX C: UCPRC MB-DENSE MIX DESIGN SUMMARY

- Table C.1: MB-15 Stability Values
- Table C.2:
   MB-15 Air Voids Analysis
- Table C.3: MB-4 Stability Values
- Table C.4: MB-4 Air Voids Analysis
- Table C.5: MAC-15 Stability Values
- Table C.6: MAC-15 Air Voids Analysis
- Table.C.7: OBC Summary
- Figure C.1: Stability vs. Percent Asphalt Content
- Figure C.2: Air Voids vs. Percent Asphalt Content

Binder:	<b>MB-15</b>							
Aggregate:	19 mm Max	<b>Dense Grad</b>	ed					
Specimen No.		1	2	3	4	5 <sup>a</sup>	6	7
% AC b	y Wt. Mix	4	4.5	5	5.5	6.00	6.5	7
Total Load pounds	Unit Load-P <sub>v</sub> (psi)		Stabilomet	ter Pressure G	auge Reading	s P <sub>h</sub> Horizent	al Pressure	
1000	80	11	13	12	12	13	13	14
2000	160	16	20	19	18	22	20	21
3000	240	23	26	25	24	29	27	30
4000	320	30	33	32	30	35	34	40
5000	400	36	36	35	37	42	43	50
6000	480	46	43	41	45	49	52	60
Displaceme	ent Turns-D <sub>2</sub>	3.84	3.65	3.65	3.8	3.45	3.55	3.35
Uncorrected Hveem	37	38	39	36	35	34	32	
Ht. of Specimen mm or inches		69	69	67	68	67	68	68
Corrected Hveem S	stabilometer Value-S	40	41	42	39	38	37	35

 Table C.1: MB-15 Stability Values

a) specimens showed slight flushing at 6.5% and moderate flushing at 7%

Specimen Number	% Asphalt Content	Wt. in Air(A)	Wt. in Air w/ Paraffin(D)	Wt. in water w. Paraffin(E)	Wt. in water	Bulk-G <sub>mb</sub> Sp. Gr. (Method A)	Max Specific Gravity(G <sub>mm</sub> )	Relative Density (RD)	% Air Voids
1	4	1210.4	1214.13	658.7	685.7	2.20	2.54	86.51	13.49
2	4.5	1223.2	1228.7	664.3	693	2.19	2.51	87.38	12.62
3	5	1226.2	1228.4	675.4	692	2.23	2.48	89.89	10.11
4	5.5	1237.7	1242.2	686.9	708.5	2.25	2.45	91.86	8.14
5	6	1233.2	1236.2	689.2	705.3	2.27	2.42	93.73	6.27
6	6.5	1244.7	1249.2	701.1	711.6	2.29	2.39	95.80	4.20
7	7	1255.9	1261.7	707.6	718.3	2.29	2.37	96.96	3.04

Table C.2: MB-15 Air Voids Analysis

 Table C.3: MB-4 Stability Values

Binder:	<b>MB-4</b>								
Aggregate:	19 mm	19 mm Max dense graded							
Specimen									
No.		1	2	3	4	5ª	6	7	
% AC by	Wt. Mix	4	4.5	5	5.5	6.00	6.5	7	
Total Load	Unit Load-								
pounds	P <sub>v</sub> (psi)		Stabil	ometer Pressure	Gauge Reading	gs P <sub>h</sub> Horizental	Pressure		
1000	80	12	11	12	12	11	12	16	
2000	160	17	15	17	18	17	18	24	
3000	240	21	19	21	25	23	25	34	
4000	320	26	22	26	31	30	34	46	
5000	400	31	29	33	39	38	44	59	
6000	480	38	36	41	47	47	56	76	
Displaceme	ent Turns-D <sub>2</sub>	3.5	3.94	3.72	3.25	3.25	3.32	3.16	
Uncorrect	ed Hveem								
Stabilomet	ter Value-S	43	42	40	39	39	35	29	
Ht. of Spec	imen mm or								
inches 67 69				69	68	69	68	68	
Correcte	d Hyeem								
Stabilomet	ter Value-S	46	45	43	42	42	38	32	

a) specimens showed slight flushing at 6.5% and moderate flushing at 7%

Specimen Number	% Asphalt Content	Wt. in Air(A)	Wt. in Air w/ Paraffilm(D)	Wt. in water w. Paraffilm(E)	Wt. in water	Bulk-G <sub>mb</sub> Sp. Gr. (Method A)	Max Specific Gravity(G <sub>mm</sub> )	Relative Density (RD)	% Air Voids
1	4	1219.1	1222.6	670.8	687.8	2.22	2.54	87.67	12.33
2	4.5	1228.6	1231.5	665	687.8	2.18	2.51	86.99	13.01
3	5	1235.9	1239.4	677.1	699.4	2.21	2.48	89.33	10.67
4	5.5	1247.4	1250.3	696.5	713.4	2.27	2.45	92.53	7.47
5	6	1249	1254.3	693.1	713	2.25	2.42	92.94	7.06
6	6.5	1251.2	1253.8	701	714.7	2.28	2.39	95.11	4.89
7	7	1252.8	1256	708.1	717.3	2.30	2.37	97.31	2.69

 Table C.4:
 MB-4 Air Voids Analysis

 Table C.5: MAC-15 Stability Values

Binder:	MAC-15							
Aggregate:	<b>19 mm</b>	Max Dense Gra	aded				•	•
Specimen No.		1	2	3	4	5 <sup>a</sup>	6	7
% AC b	y Wt. Mix	4	4.5	5	5.5	6.00	6.5	7
Total Load pounds	Unit Load- P <sub>v</sub> (psi)		Stal	bilometer Pressur	e Gauge Reading	s P <sub>h</sub> Horizental P	ressure	
1000	80	11	10	11	12	13	19	16
2000	160	15	14	16	15	20	27	24
3000	240	19	17	21	22	27	36	34
4000	320	23	22	28	29	35	45	44
5000	400	28	28	36	37	45	53	55
6000	480	35	34	44	46	54	64	68
Displaceme	ent Turns-D <sub>2</sub>	3.05	3.44	3.45	3.4	3.25	3.1	3.12
Uncorrected Hv Val	eem Stabilometer lue-S	49	46	39	39	35	32	31
Ht. of Specime	en mm or inches	67	69	69	68	68	67	68
Corrected Hve Val	prrected Hveem Stabilometer Value-S 52 49 42 42 38 35						35	34

a) specimens showed slight flushing at 6.5% and moderate flushing at 7%

Specimen Number	% Asphalt Content	Wt. in Air(A)	Wt. in Air w/ Paraffin(D)	Wt. in water w. Paraffin(E)	Wt. in water	Bulk-G <sub>mb</sub> Sp. Gr. (Method A)	Max Specific Gravity(G <sub>mm</sub> )	Relative Density (RD)	% Air Voids
1	4	1227.1	1232.9	673.1	694.7	2.22	2.54	87.38	12.62
2	4.5	1242	1245	684	703	2.23	2.51	88.82	11.18
3	5	1242.6	1249.9	683.8	706.3	2.23	2.48	89.88	10.12
4	5.5	1243.7	1249.4	685.2	710.1	2.23	2.45	91.05	8.95
5	6	1244.4	1253.9	700.8	714.3	2.29	2.42	94.78	5.22
6	6.5	1241.6	1246.9	697.2	710.3	2.28	2.39	95.44	4.56
7	7	1248.3	1255.8	705.5	717.9	2.30	2.37	97.39	2.61

Table C.6: MAC-15 Air Voids Analysis

 Table C.7: OBC Summary

Aggregate:	19 mm dense-graded								
	Recommended OBC Range (%)								
Mix	Lower Limit	Upper Limit	Design						
MB-4	6.3	6.6	6.3						
MB-15	6.2	6.5	6.2						
MAC-15	5.9	6.0	6.0						



Figure C.1: Stability vs. Percent Asphalt Content



Figure C.2: Air Voids vs. Percent Asphalt Content